

INTERNATIONAL FOOD POLICY RESEARCH INSTITUTE



INITIATIVE ON Rethinking Food Markets

IFPRI Discussion Paper 02168

January 2023

The Role of Market Concentration in the Agrifood Industry

Manuel A. Hernández

Alvaro Espinoza

Maria Lucia Berrospi

Koen Deconinck

Johan Swinnen

Rob Vos

Markets, Trade, and Institutions Division Director's General Office

INTERNATIONAL FOOD POLICY RESEARCH INSTITUTE

The International Food Policy Research Institute (IFPRI), a CGIAR Research Center established in 1975, provides research-based policy solutions to sustainably reduce poverty and end hunger and malnutrition. IFPRI's strategic research aims to foster a climate-resilient and sustainable food supply; promote healthy diets and nutrition for all; build inclusive and efficient markets, trade systems, and food industries; transform agricultural and rural economies; and strengthen institutions and governance. Gender is integrated in all the Institute's work. Partnerships, communications, capacity strengthening, and data and knowledge management are essential components to translate IFPRI's research from action to impact. The Institute's regional and country programs play a critical role in responding to demand for food policy research and in delivering holistic support for country-led development. IFPRI collaborates with partners around the world.

AUTHORS

Manuel A. Hernandez (<u>m.a.hernandez@cgiar.org</u>) is a Senior Research Fellow in the Markets, Trade, and Institutions Division of the International Food Policy Research Institute (IFPRI), Washington, DC.

Alvaro Espinoza (alvaro.espinozah@gmail.com), is a Consultant for IFPRI, Santiago de Chile, Chile.

Maria Lucia Berrospi (<u>m.l.berrospi@cgiar.org</u>) is a Research Assistant in IFPRI's Markets, Trade, and Institutions Division, Washington, DC.

Koen Deconinck (<u>Koen.deconinck@oecd.org</u>) is an Economist/Policy Analyst at the Organisation for Economic Co-operation and Development (OECD), Paris, France.

Johan Swinnen (j.swinnen@cgiar.org) is IFPRI's Director General and Managing Director, Systems Transformation, CGIAR, Washington, DC.

Rob Vos (<u>r.vos@cgiar.org</u>) is Division Director of IFPRI's Markets, Trade, and Institutions Division, Washington, DC.

Notices

¹ IFPRI Discussion Papers contain preliminary material and research results and are circulated in order to stimulate discussion and critical comment. They have not been subject to a formal external review via IFPRI's Publications Review Committee. Any opinions stated herein are those of the author(s) and are not necessarily representative of or endorsed by IFPRI.

² The boundaries and names shown and the designations used on the map(s) herein do not imply official endorsement or acceptance by the International Food Policy Research Institute (IFPRI) or its partners and contributors.

³ Copyright remains with the authors. The authors are free to proceed, without further IFPRI permission, to publish this paper, or any revised version of it, in outlets such as journals, books, and other publications.

Table of Contents

Abstractir
Acknowledgement(s)
Summaryv
1. Introduction
2. What is the current state of market concentration across the agri-food industry?
2.1 Upstream
Fertilizers
Agrochemicals
Seeds
2.2 Midstream
2.3 Downstream
3. Market concentration, conduct and performance in agri-food systems
3.1 Upstream
Fertilizers
Seeds
3.2 Midstream
3.3 Downstream
4. Implications of market concentration for food security and nutrition and environmental sustainability
4.1 Implications for food security and nutrition
4.2 Implications for environmental sustainability4
5. Concluding remarks and the way forward
References

Abstract

The role of market concentration and potential market power exertion in the agri-food industry is a topic of longstanding interest and concern to policymakers, stakeholders, and researchers. This study provides a comprehensive overview of recent trends in market concentration upstream, midstream, and downstream the agri-food industry at the global, regional, and country level, and assesses how and to what extent concentration could be affecting market conduct and performance of food systems in developed and developing countries. The analysis additionally discusses, to the extent detectable, implications of concentration, including vertical and horizontal integration that favor concentration, for food security and nutrition and environmental sustainability. While market concentration in the agri-food industry has increased across most segments, the evidence on market power exertion is inconclusive. Several knowledge and data gaps are identified and additional research is necessary to derive more general conclusions and policy recommendations.

Acknowledgement(s)

This paper was prepared for the Food Systems Economic Commission (FSEC). Funding from the EAT Foundation is gratefully acknowledged. The authors are grateful to Caterina Ruggieri, Sarah Lowder and Rachel Nugent for helpful comments on earlier drafts of this paper, as well as to Ravi Kanbur and other members of the FSEC for their comments and insights provided during presentations of the paper's findings to the Commission. This research further feeds into the new One CGIAR research program on Rethinking Food Markets and Value Chains for Inclusion and Sustainability. Other CGIAR centers participating in Rethinking Food Markets include: Alliance Bioversity & CIAT, the International Maize and Wheat Improvement Center (CIMMYT), the International Center for Agricultural Research in the Dry Areas (ICARDA), the International Institute of Tropical Agriculture (IITA), the International Water Management Institute (IWMI), and WorldFish. We thank all funders who support this research through their contributions to the CGIAR Trust Fund: https://www.cgiar.org/funders/.

Summary

The role of market concentration and potential market power exertion in the agri-food industry is a topic of longstanding interest and concern to policymakers, stakeholders, and researchers. This study provides a comprehensive overview of recent trends in market concentration upstream, midstream, and downstream the agri-food industry at the global, regional, and country level, and assesses how and to what extent concentration could be affecting market conduct and performance of food systems in developed and developing countries. The analysis similarly discusses, to the extent detectable, implications of concentration, including vertical and horizontal integration that favor concentration, for food security and nutrition and environmental sustainability. While market concentration in the agri-food industry has increased across most segments, the evidence on market power exertion is inconclusive. Several knowledge and data gaps are identified, and additional research is necessary to derive more generalized conclusions and policy recommendations.

What is the current state of market concentration in the agri-food industry?

- ✓ Market concentration in the agri-food industry (from agricultural inputs through the retail sector) has generally increased over the past years.
- ✓ The upstream segment of the agri-food chain is composed of highly concentrated industries, including fertilizers, agrochemicals, and seeds, which show trends of consolidation (i.e., reduction in the number of firms in a market) with important global market players.
- ✓ Several upstream firms have consolidated (benefitting from economies of scale, i.e., saving in unit costs by producing in larger scale) and have presence in multiple countries with subsidiary firms. Some of them further operate their exports through common trading companies, which contribute to concentration in the distribution channel.
- ✓ Markets midstream the chain, particularly food manufacturing and processing industries (that are more developed and consolidated in high income countries while scarce in low income ones), generally show moderate to high levels of concentration that have either increased or remained stable in recent years.
- ✓ The few midstream industries that show low levels of concentration include grain mill and processed meat products, although it is not a general pattern across all major producing countries in each region. For instance, Australia and the US show relatively high levels of

concentration for grain mill products that include breakfast cereals, while Australia and Brazil do the same for processed meats.

- ✓ Downstream, grocery retailers and especially supermarkets, show relatively high concentration levels both in developed and less developed regions, where indicators in the latter can also be moderate to high (or even higher than in developed regions) as is the case of South Africa, Brazil, or Mexico. Similarly, large multinational firms have presence across different regions and exhibit growth and consolidation through acquisitions of local firms.
- ✓ Grain commodity trade is also highly concentrated. There are relatively few companies (trader firms) purchasing bulk agricultural commodities for international trade.
- ✓ Important data gaps remain across industries and regions, which combined with data aggregation challenges (including defining the relevant market in terms of geographical scale and product definition), limit a comprehensive picture of market concentration in agri-food systems and constitute an invitation for devoting more efforts on data access, compilation, and processing.

What factors contribute to increasing concentration in the agri-food industry?

✓ Market concentration may originate from multiple factors. Over the past years, mergers and acquisitions combined with the transformation of agri-food chains (including the supermarket revolution) have particularly favored market concentration or consolidation in input agribusiness, food processing, and retailing companies.

Is market concentration per se bad?

✓ A typical concern in the industry is that high levels of concentration of market shares among a few large sellers (buyers) may result in market power exertion, imbalance in bargaining power, or tacit collusion that could affect the efficiency of food systems. Yet, the relationship between concentration and market power exertion is complex and the evidence supporting market power abuse is not necessarily obvious and may be context specific. Concentration alone is not sufficient to exert market power or to effectively coordinate strategies among few large market players as this also depends on other market characteristics and contextual factors. ✓ Deriving conclusive evidence regarding market power exertion in concentrated markets is thus both theoretically and empirically challenging considering the multiple factors at play that may vary by context and the lack of wide-ranging and detailed data.

What is the relationship between market concentration and conduct and performance in agrifood systems?

- ✓ Studies focusing on midstream and downstream markets in developed economies, for which more data is available, do not find strong evidence of market power exertion, especially buyer power from food processing firms. In developing countries, the literature is scarcer but shows the importance of contextual elements such as remoteness, financial liquidity constraints, credibility, and reputation issues in conferring considerable market power; high transaction costs (e.g., search, storage, contracting costs) can also sharply limit competition as well as local trade frictions (e.g., transportation and other related costs that create price gaps across areas) and lack of appropriate processing facilities. In these contexts, intermediaries can capture large economic rents in detriment of consumers and farmers. The emerging experimental evidence in developing countries is, in turn, mixed, although it is not generalizable as it has largely focused on establishing the relevance of a specific market mechanism or characteristic in the value chain under study.
- ✓ Upstream, high levels of concentration in fertilizer and seed markets, combined with elevated levels of multimarket contact in those industries (where big multinational firms compete with the same rivals in multiple markets) are a matter of concern, including recent record high fertilizer prices (in a context where supply disruptions and export restrictions or bans to one of the few major worldwide producers can provoke immediate impacts on global fertilizer availability and accessibility).
- ✓ As more disaggregated data can be accessed or compiled (including detailed transaction data), new approaches are being developed to better understand the complex relationship between market concentration, conduct, and performance. More research is similarly needed to assess the impact of unfair trading practices (UTPs) along the agri-food chain (not only on farmers), while efficiency and rent distributional implications of some practices are not fully clear. In developing countries, the implications of the transformation of food value chains (including favoring concentration) on market performance could also be studied relying on advanced

empirical industrial organization methods (widely used in developed economies), as more data, including scanner data, become available.

Implications of market concentration for food security and nutrition

- ✓ In rural areas, the association between participation in modern commercialization channels, especially supermarket channels that have increased vertical integration and coordination favoring concentration along the chain, and access to better diets among small farmers in lower income countries appears to be positive, but the literature is still relatively limited to derive more generalized conclusions.
- ✓ The evidence is generally more conclusive on the positive contribution of participation in value chains of supermarkets on rural smallholders' income, consumption expenditure, labor market insertion, and agricultural productivity.
- ✓ In urban areas, the relationship between supermarkets' presence and consumer health and nutrition remains poorly understood. In both developed and developing countries, the evidence is not yet robust enough to be conclusive regarding the positive or negative effect of supermarkets on consumer nutritional status. Part of this ambiguity could be explained by varying driving forces that influence urban consumer food choices, including market competition that has not been encompassed in previous analyses of the linkages between supermarkets and dietary patterns, probably due to the lack of detailed data.

Implications of market concentration for environmental sustainability

- ✓ The literature on the links between market concentration and environmental sustainability can be studied from different perspectives as vertical and horizontal integrations that favor concentration may occur both upstream and downstream the food value chain, which can result in varying competitive practices and environmental consequences.
- ✓ Yet, the current evidence is highly context specific as it pertains to specific market segments, supply chains, or regions. In the seeds sector, for example, the evidence is mixed. Similarly, the literature on food loss and waste (FLW) points out that food waste could be a symptom of structural problems that relate to power imbalances and lack of incentives to minimize food waste.

✓ There is an important research gap that could be fulfilled by studying potential environmental effects across different value chains and market segments that are important for the food industry, as well as on the efficient use of resources such as water, land, and energy. The scarce literature in developing countries is also evident where the lack of concentration (or vertical integration) could alternatively be a cause of food waste as smallholder farmers typically do not have the incentives or means to adopt food loss preventive practices, while downstream coordination problems with small-scale operating agents, whose additional lack of sufficient storage, transportation, and processing capacities, could contribute to food losses.

Implications for research

✓ While increasing levels of concentration through horizontal and vertical integrations can be associated with higher efficiencies (especially through economies of scale and better coordination), they could also result in anticompetitive behavior in the absence of these efficiencies and other offsetting factors (such as technological efficiencies or pro-competitive gains). Still, the occurrence of mark-up pricing, unfair trading practices, collusion, crowding out of small-scale businesses, lower access to healthy diets, less innovation, and more (or less) food waste may be context-specific and influenced by factors other than concentration. As pointed out above, several key data and research gaps remain, to better understand these relationships and derive more generalized conclusions.

Implications for policy

✓ It is important to invest in stronger national and multinational competition authorities and encourage more coordination among them to facilitate the convergence and effective enforcement of antitrust laws at the local, regional, and global level, and ensure that regulations and other policies do not accidentally create barriers to entry, restrain competition, or restrict trade between countries. Yet, it is equally important to keep in mind that guaranteeing competition among agri-food markets will not solve poverty, food insecurity and malnutrition, and environmental issues, such that other, more targeted policies are additionally needed.

- ✓ There is also an important need to expand capabilities of competition authorities to perform more in-depth market studies in specific industries or sectors along the agri-food value chain and encourage the disclosure of more detailed (and frequent) data.
- ✓ With the wide use of new technologies of information, the development of easily accessible virtual platforms (whether through local, regional, or global initiatives) that report relevant market data, promote business (trade) opportunities and access to new markets, or even permit to submit complaints about potential regulatory violations or unfair trading practices, could contribute to increase market transparency and efficiency across different agri-food industries.

1. Introduction

Market structure, defined as the number and type of sellers (buyers) in a market and the nature and level of competition between them, plays a key role in the functioning of agricultural and food markets and can have important implications for prices, product quality and innovation, contractual arrangements along the supply chain, income generation, and welfare distribution. Of particular interest and concern to policymakers, stakeholders, and academics is the role of market concentration and potential market power exertion in agri-food supply systems (IPES-Food, 2017; MacDonald, 2017; Saitone and Sexton, 2017; Sheldon, 2017; Bonanno et al., 2018; Swinnen, 2020; Deconinck, 2021). A typical concern is that high levels of concentration of market shares among a few large sellers (buyers) may result in market power abuse, imbalance in bargaining power, and/or tacit collusion that could affect the efficiency of food systems where these large actors capture most of the rents and alter welfare distribution, especially against poor smallholder producers (and consumers). This raises additional concerns about the implications of market concentration for food security and nutrition and overall environmental sustainability.

However, the relationship between concentration and market power exertion is complex and the evidence supporting market power abuse is not necessarily obvious and may be context specific (Dillon and Dambro, 2017; Lloyd, 2017; Sheldon, 2017; Sheldon, 2018; Swinnen, 2020; Deconinck, 2021). In particular, market concentration may originate from multiple factors. These factors include natural barriers to entry resulting from reduced market size or limited availability of resources and high up-front required investments (e.g., agricultural input markets); natural growth and changing trends among traders, processors, and retailers towards horizontal and vertical integration (e.g., mergers and acquisitions, the "supermarket revolution", contract farming) allowing them to capture larger shares of a market or favoring market consolidation (i.e., a reduction in the number of firms involved in a market); overall changes in consumer demand favoring product differentiation and monopolistic competition (i.e., many sellers competing against each other but offering differentiated products by branding or quality); and policies and regulations (or deregulations) that could restrict market entry or favor concentration among a few major players along the supply and distribution channels of agricultural and food value chains

(e.g., market deregulation in Asia in the 1980s and 1990s that promoted the entry of foreign large companies and reduced space for small local players).

In the case of input agribusiness, food processing, and retailing companies, market concentration or consolidation has occurred through natural growth as well as through mergers and acquisitions (Dobson et al., 2003; Swinnen, 2020; Deconinck, 2019, 2020). On this matter, Barrett et al. (2019, 2021) describe how the transformation of agri-food value chains in low- and middle-income countries occurs from traditional to modern markets and how this relates to concentration. The authors identify three stages of transition: a first stage characterized by traditional spot markets, which are typically competitive with many agents competing on price, volume, and observable quality terms; an intermediate stage with increased inter-firm competition both upstream (commodity procurement) and downstream (retailing), increasing firm pressure to reduce costs and coordinate with suppliers to provide feedstocks and differentiated products that are more highly valued in the target market (Swinnen, 2020); and a modern stage with re-concentration downstream, particularly in the more capital-intensive segments, among a shrinking number of large firms that have become ever more important in the value chain (Swinnen and Vandeplas, 2010). Besides natural barriers to entry, re-concentration can similarly emerge upstream if intellectual property rights confer market power as in the global seed industry today (OECD, 2018a; Deconinck, 2020).

While mergers and acquisitions combined with the transformation of agri-food chains over the past years have favored concentration in the industry, it is also important to keep in mind that concentration *per se* is not inevitably bad and documenting and deriving conclusive theoretical and empirical evidence regarding market power exertion is challenging due to multiple factors at play and lack of wide-ranging and detailed data (beyond prices). Although market concentration (or consolidation) may facilitate coordination between fewer market players (Fraas and Greer, 1977; Baker and Farrell, 2019), concentration alone is not sufficient for firms to be able to exert market power or to effectively coordinate strategies with other actors, or for them to (tacitly) collude as this also depends on other market characteristics and factors, including the type and nature of competition.

This paper provides a comprehensive assessment of the role of market concentration in the agrifood industry. The study reviews the available evidence on market structure (concentration), conduct (behavior), and performance (efficiency and rent distribution) across different segments of the agri-food supply chain and identifies key aspects of agri-food chains that should be considered when examining market concentration and market power relationships. After first documenting global and regional concentration patterns at different segments of the chain, the analysis then focuses on assessing whether market concentration translates into different manifestations of market power, considering the nature of competition between market actors and using concrete examples available in the literature. The discussion on market conduct focuses on an examination whether and how concentration (i) translates into lower buyer prices (i.e., buyer power) or higher seller prices (i.e., seller power), (ii) limits product quality and/or innovation, (iii) promotes unfair trading practices or contracts, or (iv) results in coordination strategies among powerful actors or tacit collusion. The discussion on performance, in turn, focuses on potential impacts of concentration on price margins (markups), profits, and rent allocation. Evidence on the implications for food security and nutrition, including access to healthy diets, and environmental sustainability, including food loss and waste, are also discussed.

Figure 1.1 depicts the segments of the agri-food industry considered for this study. Upstream, these comprise key agricultural input markets (e.g., fertilizers, agrochemicals, seeds), which are highly concentrated global markets that seem to be in a consolidation path; midstream, agricultural production (farming), storage, distribution, and processing that traditionally have shown low to medium levels of concentration, except for food processing and manufacturing where concentration is increasing; and wholesale, retail, and food services downstream that generally exhibit medium to low concentrations, with the exception of retailing (especially supermarket retailing) where concentration is higher and increasing.

Two standard measures of market concentration are (i) the four-firm concentration ratio (CR4), which is the sum of the market shares of the four largest firms competing in a market, and (ii) the Herfindahl-Hirschman index (HHI), which is the sum of the squared of the market share of each

firm operating in a market.¹ To facilitate comparison of concentration patterns across different segments of the agri-food chain and across regions (countries), we focus on concentration ratios, as more data is available allowing to calculate these ratios. In the same vein, we treat these measures as indicators of market structures or the level of concentration in a market, but not necessarily as indicators of the degree of competition or certain market power that requires a more detailed analysis, as extensively discussed in the paper.

Overall, the study aims to inform the Food System Economics Commission on recent trends in market concentration in the agri-food industry and on how and to what extent concentration could be affecting market conduct and performance of food systems, as well as implications for food security and nutrition and environmental sustainability. On this basis, the study provides recommendations for the research and policy agenda and signals knowledge and data gaps to be addressed through further analyses.

The remainder of the paper is organized as follows. Section 2 presents an overview of market concentration across different segments of the agri-food industry based on the most recent available data. Section 3 reviews the existing empirical evidence regarding how and the extent to which market concentration could affect conduct and performance in agri-food systems, while Section 4 discusses the implications of concentration for food security and nutrition and environmental sustainability. Section 5 provides concluding remarks and recommendations moving forward.

¹ The HHI ranges from 0 to 10,000 (e.g., in a market with two firms and shares 60% and 40%, the HHI is 60x60 + 40x40 = 5,200). The main difference between the HHI and CR4 is that the HHI puts more weight on larger firms. All else equal, a reduction in the number of firms in a market leads to a higher HHI, however, this would not occur with the CR4 if the shares of the four largest firms remain the same. The simplicity of the CR4 could thus be a limitation in oligopoly markets with a larger number of firms. In any case, both measures are well accepted in analysis of market concentration, but for its properties the HHI plays a more important role in the assessment of mergers.

2. What is the current state of market concentration across the agri-food industry?

This section presents an overview of market concentration across the agri-food industry. Concentration measures and recent trends are separately described upstream, midstream, and downstream the chain at the global, regional, or country level depending on data availability. Most of the analysis is based on concentration ratios, especially the widely used four-firm concentration ratio (CR4) that is equivalent to the market share of four largest firms in any given market, but measures of the Herfindahl-Hirschman index (HHI), which is the sum of the squared market shares of all firms operating in a market, are also presented when available.

As noted earlier, the measures described in this section should be treated only as indicators of the degree of concentration in a market.² They do not necessarily represent indicators of market power, although the level and variations of these indicators, combined with the number of competitors in a specific market, are employed by competition authorities and economists to assess competitive conditions (Kwoka, 2016; Nocke and Whinston, 2022). In the case of the CR4, a market may no longer be considered competitive when four firms have more than 40% of the market share (Shepherd and Shepherd, 2004; Howard, 2016). In the case of the HHI, the US Department of Justice (DOJ) Horizontal Merger Guidelines (2010) classify industries as un-concentrated for HHI less than 1,500, moderately concentrated for HHI between 1,500 and 2,500, and highly concentrated for HHI above 2,500. Typically, industries that are classified as unconcentrated are not subject to DOJ scrutiny, while mergers that would increase the HHI by 100 points or more among moderately concentrated industries can raise competition concerns and are often evaluated; increases by 200 or more among highly concentrated industries are, in turn, presumed to be likely to enhance market power. The European Commission also has horizontal merger guidelines that adopt similar thresholds based on the HHI level and change (Nocke and Whinston, 2022).

It is worth to mention that when applied to studying market competition, the use of indicators for the degree of concentration should always be accompanied by the definition of the relevant market,

² The causes of market concentration may also be multiple (as noted in the previous section) and vary across market segments and locations, which requires a more in-depth assessment on a case-by-case basis that is beyond the scope of this study.

which varies depending on the industry considered.³ As pointed by Crespi and MacDonald (2022), easily available or published measures of market concentration often fail to fully capture economic markets and should be taken as a starting point. Even with disaggregated data, industries may not necessarily correspond to relevant economic markets, either because of too narrow or too broad definitions. In the context of agri-food markets, characteristics like the perishability of some products, the difficulty to transport livestock or the local nature of food retail and grocery stores, difficult defining the geographic scope of a market and is important to be aware of these limitations when analyzing published concentration data.

2.1 Upstream

Fertilizers

The production of fertilizer, as well as other agricultural inputs discussed below, is highly concentrated among a few countries. This is mainly explained by its intense requirements of specific raw materials and the provision of natural gas that are not available worldwide, as well as by economies of scale in production, which generate cost efficiencies for firms (i.e., lower their per-unit costs of production). Several of the major producers are also high-income countries with high trade volumes and comparative advantage in commercialization at the global level.

Using data from International Fertilizer Development Center (IFDC) for 2008-2009, Hernández and Torero (2011, 2013) show that most of the global production capacity for the main fertilizers (ammonia, urea, ammonium nitrate, monoammonium phosphate, MAP; diammonium phosphate, DAP; and phosphoric acid, potash, and nitrogen-phosphate-potassium, NPK, compounds) is concentrated in few countries. Except for ammonium nitrate, the top five countries control more than 50% of the world's production capacity for all these fertilizer products. In the case of potash, Canada and Russia alone are responsible for more than half of the production capacity, while in

³ In product markets, the relevant market (and the relevant industry) should include all close substitutes and exclude those who are not. In input markets, the relevant market definition must include all competing buyers of a product.

the case of for ammonia, urea, DAP/MAP, and phosphoric acid, China, India, Russia, and the US are among the countries that dominate their production capacity.⁴

Hernández and Torero (2011, 2013) also show a high level of concentration among producers within each main producing country (except for China), which is in part explained by the large upfront investments required that result in economies of scale in production (Kim et al., 2002). In most cases, the top four firms control more than half of each country's production capacity. Concentration of potash production at the country level is the most extreme case: in four of the five main potash-producing countries, the CR4 index is 100%. For DAP/MAP and NPK, four of the five main producing countries exhibit a CR4 over 60%, while for urea three countries show a CR4 over 50%.

Market research data compiled from Euromonitor International (hereafter Euromonitor) permit to depict more recent concentration trends in the fertilizer industry. Using information on company shares for annual production turnover of fertilizers and nitrogen compounds at current prices (in US dollars applying year-over-year exchange rates), market concentration indicators based on country shares in the global market can be calculated by simply adding company shares within each country.⁵ According to these calculations, a global CR4 index of 56% is obtained for 2012 (compared to 43% in 1997) with China being the main producer (38%), followed by Russia (6%), India (6%), and Brazil (6%). The global level of concentration peaked in 2015 to 65% and has returned to 56% in 2021 where China (40%), Russia (6%), India (6%), and Brazil (4%) remain as the main producers. Hence, these high levels of concentration at the global level have maintained over the past years with the consolidation of China as the main producer.

At the country level, the CR4 indexes among the top four producing countries have similarly either maintained or slightly increased between 2012 and 2021. As shown in **Figure 2.1**, in India and

⁴ The production of phosphate- and potassium-based fertilizers is generally more concentrated than nitrogen-based fertilizers as the availability of phosphate rock and potash deposits is more geographically limited.

⁵ Fertilizers and nitrogen compounds comprise nitrogen products associated with fertilizer production (nitric and sulphonitric acids, ammonia, ammonium chloride, ammonium carbonate, nitrites, and nitrates of potassium) and phosphate, potash, and compound fertilizers.

Russia the CR4 index is above 40%, while in Brazil and China it increased from 24.5% and 11.5% to 27.6% and 14%, respectively. The same companies also remain among the top producers in each country and in the same ranking; the only exception is Yunnan Yuntianhua Co Ltd in China that was not among the top four producers in 2012 and became the most important producer in 2021 with a 4.2% of the total volume of sales.

Overall, the fertilizer industry is in a consolidation path with major companies holding important market shares, such as Yara International (Norway) and The Mosaic Company (US) and the recent merger of PotashCorp with Agrium (Canada), combined with a growing importance of manufacturers in the Asia-Pacific region, particularly in China and India. Given the capital-intensive nature of the fertilizer industry, several firms have consolidated to benefit from economies of scale and the size of dominant firms inhibit the entry of smaller companies (IPES-Food, 2017). Several companies also have presence in multiple countries, including subsidiary firms in developing countries, due to cost reductions in gathering or producing inputs (e.g., mining) or for strategical reasons to locate near demand (e.g., Yara International and The Mosaic Company). Similarly, as noted by Fuglie et al. (2011), some of them operate their exports through common trading companies (e.g., Canpotex for PotashCorp, Agrium, and The Mosaic Company and Belarus Potash Company for Uralkali in Russia and Belaruskali in Belarus) that contributes to concentration in the distribution channel and could favor market power exertion.

Agrochemicals

The agrochemical industry also exhibits an important level of concentration. Using firm-level sales data for crop protection chemicals, Fuglie et al. (2011) notes that the global CR4 index in this industry increased from almost 29% in 1994 to 53% in 2009, which, according to this study, is comparable to the trends shown by other agricultural input industries that invest heavily in agricultural research (i.e., seeds, animal health, farm machinery). IPES-Food (2017) further points that the agrochemical industry has been the most recognizable example of consolidation in the agri-food system in recent years, with some of the largest deals in mergers and acquisitions history (both within this industry and expanding to the seed sector), which have additionally renewed a public debate on its implications. Based on 2014 data from the Action Group on Erosion, Technology, and Concentration (ETC), the report remarks that the world's largest agrochemical

corporations at that time, Syngenta (Switzerland), Bayer (Germany), BASF (Germany), DuPont (USA), Monsanto (USA), and Dow (USA), known as the 'Big Six', are linked to the proprietary seed industry, and controlled both 75% of the global pesticides market and 60% of the global seed market.⁶

Similar to fertilizers, market information provided by Euromonitor for the production of pesticides and other agrochemical products permits to compute market concentration measures at the global and country level and visualize more recent trends.⁷ Compared to a global CR4 index of 52% in 1997 with the US being the main producer (27%) followed by Japan (9%), France (8%), and China (8%), in 2012 the concentration ratio increased to 64% with a substantial growth from China that concentrated 31% of the global production, followed by the US (16%), Brazil (10%), and Germany (7%). The level of concentration then peaked in 2015 with a global CR4 of 68% and has remained at 63% in 2021 where China (34%), US (14%), Brazil (8%), and Germany (7%) continue to be main producers.

Within each of the four top producing countries, however, the level of concentration has followed different patterns. **Figure 2.2** shows that the CR4 index in China basically doubled between 2012 and 2021 (from 18% to 35.8%), where Wuhai He Ye Chemical Engineering Co Ltd has continued to consolidate its position as the major market player. In the US and Germany, in contrast, the

⁶ Seeds became logical and complementary acquisitions for pharmaceutical and chemical corporations investing in biotechnology. Multinational firms – including Bayer and BASF – continue to support cross-sector research in pharmaceutical (human and livestock), chemical and agrochemical applications, while others have been divesting their agricultural units. For example, Pharmacia spun-off Monsanto and AstraZeneca and Novartis spun-off their agricultural unit, Syngenta (IPES-Food, 2017). These interactions with crop protection chemicals are also noted by OECD (2018a) and Deconinck (2020), indicating that seeds and crop protection chemicals can be complements (as with herbicide-tolerance traits and herbicides) or substitutes (as with insect-resistance traits and insecticides), which can have different implications for firms' strategies (Just and Hueth, 1993). Thus, mergers are an effective way of internalizing positive spillovers from investing in complementary products and can facilitate coordination of R&D (Deconinck, 2020). The relative importance of these complementary and substitution effects is therefore essential in order to evaluate the welfare effect of a merger (OECD, 2018a).

⁷ Pesticides and other agrochemical reported comprise insecticides and herbicides, plant growth regulators and disinfectants (anti-sprouting products), and fungicides and rodenticides. While the data reported by Euromonitor cannot be disaggregated by each specific sector, it is worth noting that the off-patent herbicide sector has significantly grown over the past years, particularly in developing countries (see also Haggblade et al., 2017).

concentration ratio decreased but still remain over 40%, while in Brazil it has remained around 9-10%.

Seeds

Similar to other agricultural input industries, the seed industry is highly concentrated. Fuglie et al. (2011) indicate that the combined market share of the four largest firms in the global seeds market increased from 21% in 1994 to 54% in 2009, based on firm sales data and an estimate of the overall size of the global seed market.⁸ IPES-Food (2017), in turn, report a global CR4 index of 58% in 2011 using ETC Group sales data, which further increased to 65% in 2014, where Bayer CropScience-Monsanto (Germany-US) explains 30.1% of the global sales followed by DuPont-Dow (US), Syngenta (Switzerland), and Vilmorin & Cie (France) with 22.7%, 7.8%, and 4.4%, respectively.⁹

A more recent study on seed market concentration by the OECD (2018a) provides a comprehensive coverage for different seeds disaggregated at the country level. The estimates are based on the overall seed market value, including both commercial and farm-saved seeds (that are more common in developing countries), using data from Kleffmann amis® AgriGlobe® database that covers 65% of the global market value in maize (32 countries), 86% in soybeans (7 countries), 62% in rapeseed (17 countries), and 75% in sunflower (11 countries). Other crops considered, although with a minor coverage, include sugar beet with 43% of the global market value (4 countries), cotton with 35% (3 countries), wheat and barley with 23% (15 countries), and potato with 4% (2 countries).

In the case of maize, the CR4 index is above 60% in 30 of the 32 countries covered and above 40% in the other two countries, while the HHI is above 1,500 in 27 of them; if the indicators are based on volume instead of value, CR4 is over 60% in 28 countries and HHI is above 1,500 in 24

⁸ Besides agrochemicals and seeds, Fuglie et al. (2011) examined global market concentration over a 15year period (1994 through 2009) in animal pharmaceuticals, animal genetics, and farm machinery and find that the largest four firms in these other industries also accounted for more than 50% of global market sales by 2009.

⁹ IPES-Food (2017) similarly report high or increasing concentration levels in animal pharmaceuticals, animal genetics, and farm machinery.

countries. For soybeans, the CR4 index is 42% in Ukraine while in the other six countries is above 60% in terms of both value and volume; the HHI is similarly over 1,500 in six of them, while in four countries is above 2,500. For rapeseed, CR4 is above 60% in terms of value in 16 of the 17 countries and above 40% in the remaining country, while the HHI is over 1,500 in 12 countries; in terms of volume, CR4 is above 60% in 12 countries and the HHI above 1,500 in ten countries. Lastly, for sunflower the CR4 index is above 60% across all countries in terms of value and only one country (Russia) has a CR4 below 60% in terms of volume (58%); the HHI in this market exhibits values over 2,500 across most countries.

The OECD (2018a) study also mentions that shares based on volume could be a more representative estimate of market concentration at the global level than shares based on sale values, such as Fuglie et al. (2011) and IPES-Food (2017), because sale shares assign a greater weight to markets with higher seed prices. It is thus difficult to derive a precise estimate of global market concentration as data, when available, varies across countries and market segments that prevent aggregation. In the same vein, Bonny (2014, 2017) argues that value estimates in the seed industry tend to vary across different sources and methods, sales of small and medium-sized firms are often underestimated, and farm saved seeds are not necessarily always included in the estimations; as a result, concentration measures based on under-estimates of total market size could overstate the degree of market concentration. Deconinck (2020) provides additional elements regarding potential shortcomings of global concentration calculations. The author remarks that GMO seeds have much higher unit prices such that these markets could be over-weighted in global estimations (and thereby Monsanto's market share); similarly, a large portion of global seed markets is barely commercialized (e.g., Sub-Saharan Africa, India) and seeds are very context-specific (unlike, for instance, pesticides), so local instead of global market shares can offer more useful statistics.

Bryant et al. (2016) also analyzes the level of concentration for maize, soybean, and cotton seed markets in the US using 2014 and 2015 corporate level data from multiple data sources.¹⁰ Their estimates reveal a very high level of concentration across all three seed markets. The CR4 index

¹⁰ Maize and soybean 2014 data is obtained from Begemann, S. (2015), while cotton 2015 data is obtained from the US Department of Agriculture, Agricultural Marketing Service 2015 Report for "Cotton Varieties Planted" (AMS-USDA, 2015).

is 83% for maize where Monsanto alone has a 35.5% share and DuPont Pioneer another 34.5%. In soybeans, the CR4 is 76.2% where DuPont Pioneer and Monsanto have shares of 33% and 28%, respectively; while in cotton (upland) the CR4 is 91.4% where Bayer has a market share of 38.5% followed by Monsanto with a 31.2% share. These estimates, however, reflect the situation before the merger of Bayer and Monsanto, which also required these companies to divest certain businesses to get the merger approved. Deconinck (2020), using various data sources, elaborate market shares in US upland cotton seed for the period 1970–2017, and notes that publicly available data for this crop makes it possible to track in detail the evolution of market shares over time;¹¹ based on the author compilation, the CR4 of upland cotton seed in the US has been over 90% since 2000.

2.2 Midstream

Despite the widespread use of market concentration statistics across several industries, reliable data midstream the agri-food chain is relatively scarce, including food manufacturing and processing. For the US, for example, nation-wide concentration ratios for food manufacturing industries are available from Economic Census data that are compiled every five years (Deconinck, 2021). Using these data, Saitone and Sexton (2017) show that, out of the 37 food manufacturing industries covered by the data in 2012, 29 were unconcentrated (HHI below 1,500); five were moderately concentrated (HHI between 1,500 and 2,500); and three (specialty canning, flavoring syrup and concentrate manufacturing, and other snack food manufacturing) were highly concentrated (HHI above 2,500). This trend is confirmed with the latest US Census data, reported in Crespi and MacDonald (2022), showing that out of 35 food processing industries with available data, only five report HHI values over 1,500 where breakfast cereal manufacturing is the most concentrated industry (2,210).¹² Yet, Saitone and Sexton (2017) point out that these nation-wide estimates using North American Industrial Classification System (NAICS) codes to identify

¹¹ The data sources include OECD (2018a), US Department of Agriculture, Agricultural Marketing Service, "Cotton Varieties Planted" various years (1982–2017), and Fernandez-Cornejo (2004).

¹² Nevertheless, Crespi and MacDonald (2022) similarly present CR4 indicators for 42 food processing industries using the US Census data and show that 23 report a CR4 above 40% and 11 above 60%; some of these industries include: other snack food manufacturing; animal (except poultry) slaughtering; specialty canning; soybean and other oilseed processing; cane sugar manufacturing; and breakfast cereal manufacturing.

industries, are not very relevant for analyzing potential buyer power in agricultural markets, an argument also acknowledged by Crespi and MacDonald (2022). As farm products are often bulky and perishable, the relevant procurement markets are local or regional. The same applies for final consumer products. Detailed information at these more disaggregated scales is generally not available. Crespi and MacDonald (2022) conclude that with this evidence, changes in recent years have been industry-specific, and that the analysis of concentration remains a data-intensive and industry-specific proposition.

Deconinck (2021), in turn, provides estimates of market concentration in the US livestock slaughter for the period 1980-2015, based on various statistical reports from the Grain Inspection, Packers and Stockyards Administration (GIPSA-USDA, 2016), and shows an increase in concentration in recent decades. In particular, the share of steers and heifers slaughtered by the four largest firms increased from less than 40% in 1980 to 85% in the last decade; for cows and bulls, from around 10% to close to 60%; and for hogs, from one-third to around two-thirds. These increasing concentration ratios have additionally raised competition concerns among the US Department of Agriculture and Department of Justice as hog and cattle producers repeatedly mentioned concentration and centralization of meatpacking as an issue (US Department of Justice, 2012). While allegations of anti-competitive conduct by meatpackers have continued, Deconinck (2021) points to empirical research showing that producers may be better off because of lower costs and innovations introduced (Wohlgenant, 2013) or even suggesting that the markets have become more competitive since the 1980s (Ji et al., 2017). As noted earlier, this discussion is connected to the fact that market structure (concentration) itself is not necessarily a good predictor of competitive conditions. This matter is further discussed in the next section.

Estimates for food manufacturing in the European Union are available from the European Commission (2014) study on the economic impact of modern retail. Data for 14 member states in 2012 across 23 product categories, show that baby food, frozen ready-cooked meals, and cereals and coffee exhibit the highest concentration levels, while ham, bread, and cheese exhibit the lowest levels. When averaging across the 23 product categories in each country, the supplier concentration

varies from a HHI less than 1,400 in Germany to over 2,800 in Denmark, with a median of 2,100. In addition, average concentration increased between 2004 and 2012 across most countries.¹³

Comparable data for other regions and countries are generally difficult to find, although for some countries or industries detailed studies exist. In Mexico, Deconinck (2021) refers to a 2015 study by the Federal Economic Competition Commission (COFECE) that reports estimates of concentration levels at various stages of the agri-food chain using economic census data. The study shows that concentration appears to vary considerably by product category as in the US and Europe. For instance, the estimated HHI is less than 1,500 for sugar cane, vegetable oils and fats, cookies and pastries, among others; moderately concentrated markets include beer, prepared foods other than frozen foods, and condiments and dressings; and highly concentrated markets includes snacks, breakfast cereals, and industrial baking. The report similarly stresses that HHI estimates based on census data may underestimate true market concentration in some markets.

In sum, publicly available estimates of market concentration midstream the food chain should be analyzed with caution as they may not capture the relevant market. This is further emphasized by Deconinck (2021) who indicates that this not only applies in terms of geographical scale (i.e., procurement markets for agricultural products will rarely parallel with national or regional market definitions used in Census data) but also for product definitions, because product categories used for concentration estimates in food manufacturing are not always the relevant ones from the farmers' perspective.

Market data from Euromonitor also permit to depict recent concentration trends for selected product categories in food manufacturing and processing that include dairy, grain milled products, and meat and meat products.¹⁴ Annual company shares are available from 2012 through 2021 for several countries based on production data (at manufacturer selling price). For illustrative

¹³ The study also provides more detailed information for selected countries based on country-specific studies.

¹⁴ Dairy products comprise butter and spreads, cheese, drinking milk products, yoghurt and sour milk products, and other dairy; grain mill products comprise grain milling and breakfast cereals; and meat and meat products comprise red meat, white meat, and red and white meat products.

purposes, the most important market (i.e., country with the largest production) is selected in each region and product category.

Figure 2.3 plots the evolution of the CR4 index for dairy processors among the selected countries. Except for Brazil and Russia, the concentration ratios are above or close to 50% and have generally remained rather stable or increased over time. China shows an 8.5-percentage points (pp) increase (from 46.9% in 2012 to 55.4% in 2021), followed by South Africa with a 7pp increase (from 53% to 60%), Australia with a 5pp increase (from 58.3% to 63.3%), and Saudi Arabia with a 2.9pp increase (from 51.5% to 54.4%). In Germany the ratio has remained rather stable at 49-50% while in the US it decreased from 58.3% to 53.6%.

Figure 2.4 depicts the evolution of the CR4 index for grain milled products. For this product category, only Australia and the US exhibit important levels of concentration. In the case of Australia, the CR4 maintained around 74-75% between 2012 and 2021 while for the US it increased from 50.9% to 56.5%. For the remaining countries (Turkey, Brazil, Russia, and China), the concentration levels are below 20% and have remained rather stable over the last decade.

Finally, **Figure 2.5** presents CR4 trends for meat and meat products. Australia also exhibits in this case the highest level of concentration, although it decreased from 67.6% in 2012 to 57.3% in 2021, followed by Brazil with a relatively stable CR4 of 43-44%. The other countries show concentration ratios below 30% that have generally maintained steady over time, except for Saudi Arabia that increased from 20.6% to 27.8%. In the US and Germany, CR4 indexes situate at 27.8% and 23.6%, respectively, in 2021, while China barely has a concentration ratio of 10.7% and Russia of 4.6%.

2.3 Downstream

Downstream the agri-food chain, most of the available information focuses on the retail sector in the US and EU, where particularly food retailing was a largely fragmented sector until the 1980s (Dobson et al., 2003). In the UK, for example, the leading five firms controlled only 8-9% of national retail goods sales in 1961 and 14.4% by 1982 (Swinnen and Vandeplas, 2010). The real

consolidation wave of the retail sector in the US and EU largely took place in the 1990's (Swinnen, 2020). In the US, the combined market share of the four largest grocery retailers increased from 14% in 1984 to 22% in 1994 and 55 % in 2001 (Swinnen and Vandeplas, 2010). Einarsson (2007) similarly reports that by 2004 in the Nordic countries the market share of the largest three retailers in the grocery market was already 79.6% in Finland, 81% in Iceland, 82% in Norway, and 91.2% in Denmark and Sweden.

More recent estimations confirm that the high level of concentration in the retail sector generally persists, which has also been a matter of concern (especially in the US and Europe). Using US Census data up to 2013, Saitone and Sexton (2017) show that the four leading grocery retailers have a combined share of close to 40% of the national market (compared to less than 20% in 1992). In several large US cities, CR4 indexes are even 60% or higher, pointing to larger concentrations at the local level. A similar trend is observed in small nonmetropolitan and rural areas, as reported by Crespi and MacDonald (2022) using data from US counties that show an increase in food retail concentration between 1990 and 2015.¹⁵ In Europe, the European Commission (2014) study on the retail sector notes that the market share of the largest five food retailers reached over 80% in the 2000s. By 2012, the HHI varied from less than 1,200 in Italy to more than 3,900 in Finland, with a median of around 2,000. Yet, in contrast with concentration among food manufacturers, average retail concentration saw a decline between 2004 and 2012 in 16 of the 26 member states for which data are available, due to growth of retailers who had a small market share in 2004 or were not even present (such as discounters), but in the other ten member states retail concentration increased (European Commission, 2014).

An up-to-date study of the retail sector in the US, performed by the Economic Research Service of the US Department of Agriculture (ERS-USDA, 2021a), remarks that the total food sales of supermarkets, other grocery (except convenience) stores, warehouse clubs, and supercenters reached 653 US billion dollars in 2019, and the concentration of sales among the largest retailers has been on an upward trend after the 2007-2009 recession. More specifically, the share of food

¹⁵ The average HHI for food retail increased from 2,071 in 1990 to 3,075 in 2015 in rural counties and from 1,617 to 2,924 in non-metropolitan small urban counties (Data from USDA, Economic Research Service and Çakır et al. (2020) using National Economic Time Series (NETS) data).

sales by the top four retailers have been continuously growing from 31% in 2012 to 34% in 2019. A contributing factor to this rising trend over the past decade has been the steady growth of Walmart Supercenters, making Walmart the largest US retailer of grocery products, followed by Kroger, Albertson's, and Target. In addition, Kroger, the largest traditional retailer, has been a major player in mergers and acquisitions activities, acquiring important retailers such as Harris Teeter and Roundy's over the past decade, while Albertsons' acquired Safeway (another important retailer) in 2015.¹⁶ Overall, more than 300 food industry mergers and acquisitions were recorded in 2019 in the US.

Available data for grocery retailers and supermarkets in Euromonitor also allows to show concentration trends between 2012 and 2021 for selected countries across the world.¹⁷ The estimates are based in this case on annual company shares using sales data (at retailer selling price). Similar to the midstream figures, the country with the largest grocery store or supermarket sales is selected in each region.

For grocery retailers, **Figure 2.6** shows important (over 40%) and somewhat increasing concentration levels among several countries. As in food manufacturing and processing, Australia reports the largest CR4 in grocery retail (from 74.4% in 2012 to 78.6% in 2021), followed by South Africa (from 47.2% to 54.3%), France (around 50%), the US (around 40%), and Russia that reached 40.2% in 2021 (exhibiting a 23.8pp increase relative to 2012). Of the remaining countries, the CR4 increased from 32.5% to 35.8% in Mexico and from 14.1% to 19% in Saudi Arabia, while in China the concentration ratio remained at 3%.

For supermarkets, there are even higher levels of concentration that in several cases have further increased over time, depicting the consolidation of this sector. As shown in Figure 2.7, In

¹⁶ Another important recent acquisition was that of Whole Foods by Amazon in 2017, although the overall market structure of the retail sector was not largely affected as Amazon was not a major player in the industry (ERS-USDA, 2021a).

¹⁷ Grocery retailers comprise retailers selling predominantly food/beverages/tobacco and other everyday groceries; these include hypermarkets, supermarkets, discounters, convenience stores, independent small grocers, chained forecourt retailers, independent forecourt retailers, food/drink/tobacco specialists, and other grocery retailers. Supermarkets comprise retail outlets selling groceries within a selling space of 400-2,500 square meters; excludes discounters, convenience stores, and independent grocery stores.

Germany, Australia, and South Africa, the CR4 index reached 96%, 89.6%, and 81.5%, respectively, in 2021 reflecting increases of 4-7pp from 2012. In the US and Saudi Arabia, CR4 levels reached around 40% in 2021, compared to 31% in 2012, while in Brazil the concentration ratio has remained relatively stable at 40-41%. Russia also shows an important increase from 12.1% to 22.8%, whereas in China the level of concentration is barely 3%.

The food service industry is another large and growing sector downstream the agri-food chain. In the US, for instance, this industry was larger than food retailing until 2019 (prior to the pandemic). According to ERS-USDA (2021b), full-service and fast-food restaurants, the two largest segments of the commercial food service market in the US, account for about 74% percent of all food service outlets in 2020. Food service companies are major customers of wholesalers that distribute specifically to this industry and do not necessarily compete with food retailers for customers. Instead, food service distributors offer value-added services designed to meet the needs of single-store restaurants and small chains. Two firms, Sysco and US Foods (formerly U.S. Foodservice), dominate this broadline distribution in the US with over 70% of the market (AMS-USDA, 2022); in 2015, Sysco's planned merger with US Foods was dropped after being blocked based on antitrust grounds.

Globally, the fast-food industry is a highly fragmented market composed of a few large and many small competitors. The market was worth 972.74 billion US dollars in 2021 and is projected to increase to 1,467.04 billion US dollars by 2028 due to a growing number and variety of quick service restaurants in developing and developed countries, increasing tourism, expanding millennial population base, and dual-working households, and despite increasing awareness of health issues related to fast food consumption (Fortune Business Insights, 2022).¹⁸ The US alone has close to one third of the global fast-food market share, while the burger and sandwich segment dominates the world market (explaining almost 36% of the market share) due to the presence of burger chains in most countries, followed by the pizza and pasta segment. According to an industry report by T4 (2021), the leading fast-food chain in the market is McDonald's with 21.4% of total revenues in 2018, followed by Starbucks with 7.5%, and KFC and Subway with 2.8% each.

¹⁸ For instance, the number of quick-service restaurants in the US increased from 285,000 in 2000 to 340,000 in 2015.

The high concentration in the international distribution and commercialization of grains also deserves some discussion. There are relatively few companies (trader firms) purchasing bulk agricultural commodities for international trade. The four major corporations that produce, process, transport, finance, and trade food and agricultural commodities (known as the ABCD): ADM - Archer Daniels Midland (USA), Bunge (USA), Cargill (USA), and Louis Dreyfus Commodities (France), are estimated to account for up to 90% of global grain trade (Murphy et al., 2012). A more recent estimate of the combined 2016 revenues of ABCD traders plus the two big emerging firms from Asia, COFCO Group (China) and Wilmar International Ltd. (Singapore), is 365 billion US dollars, a figure that far exceeds the combined global markets for seeds, agrochemicals, farm equipment, and fertilizers, according to IPES-Food (2017).¹⁹ Hence, agricultural (grain) commodity trade is highly concentrated, while much data remain proprietary and companies continue to globalize and diversify their activities.

All in all, the upstream segment of the agri-food chain is composed of highly concentrated industries, including fertilizers, agrochemicals, and seeds, that appear to be in a consolidation path with important global market players. Markets in the midstream of food supply chains, particularly food processing, as well as those further downstream, especially supermarket retailing, similarly show moderate to high levels of concentration (and consolidation in some cases) that have either increased or remained stable in recent years across several countries in different regions. The international distribution and commercialization of grain also shows high levels of concentration.²⁰ In the next section, we extensively discuss the implications of concentration on market conduct and performance in the agri-food industry, considering that this high and/or increasing concentration has been a matter of recurrent concern, particularly in the US and Europe, over the past years.

Despite the estimates and trends presented above, it is worth noting that important data gaps remain across industries and regions that prevent a more comprehensive picture of market structure in agri-food systems. IPES-Food (2017) point out difficulties in accessing detailed firm and market

¹⁹ COFCO Group is China National Cereals, Oils and Foodstuffs, a state-owned company.

²⁰ See **Table 2.1** for a summary of all figures and estimates of concentration discussed in this section.

data, which constitute a major obstacle in examining concentration of power in the agri-food industry. The study further remarks that the available analyses are typically geographically incomplete. There is, for example, a disproportionate presence of US and UK companies on global markets and emerging technologies but limited disaggregated data to understand their domestic and global roles.²¹ Lack of detailed data (including historic data) also limits understanding of the growing influence of leading companies from the Global South (e.g., China, India, Brazil). In the same vein, Fernandez-Cornejo and Just (2007) and Mammana (2014) indicate general difficulties in obtaining market share data at a disaggregated level (when examining the seed industry). These data shortcomings, combined with the issues outlined above regarding data aggregation challenges or complications, including the use of different variables for analysis (e.g., shares based on volume versus sales) or defining the relevant markets across segments in the agri-food chain, are certainly an invitation for devoting more efforts on data access, compilation, and processing.

It is fair to indicate, however, that competition authorities do not necessarily have difficulties in obtaining disaggregated market data. In many jurisdictions, these institutions can conduct market analysis on their own initiative, and in some cases even compelling testimony. Following a roundtable discussion on market studies held in 2008, the OECD and the UK launched a project in 2014-2015 to provide support to competition authorities from several countries in Latin America and the Caribbean in their use of market studies as a competition tool, and in developing a legal framework.²² Furthermore, with the release of the Market Studies Guide to Competition Authorities (OECD, 2018b) and the exchange of experiences between competition authorities, the topic has continued to trigger interest in recent years. Additional to regular functions held by the authorities, the goal of these market assessments is to verify if competition is working efficiently and identify measures to address issues detected. These measures comprise recommendations such as proposals for regulatory reform or improving information dissemination, and they can also include the opening of antitrust investigations.

²¹ Howard (2016) notes that 22 of the 40 largest food companies around the world are US- or UK-based companies.

²² Chile, Colombia, Costa Rica, Mexico, Panama, and Peru. More information about the OECD work on market studies is available at: <u>https://www.oecd.org/daf/competition/market-studies-and-competition.htm</u>.

3. Market concentration, conduct and performance in agri-food systems

Inferring a causal relationship between market structure (concentration), firm conduct, and performance is generally a difficult empirical task. Since the pioneer work of Demsetz (1973), which was later articulated as an empirical problem by Bresnahan (1989), economists have advanced from "structure-conduct-performance" studies in favor of alternative approaches, considering the necessity to correctly address problems of endogeneity, simultaneity, and causality between modeled market equilibrium outcomes (O'Brien, 2017; Berry et al., 2019; and Miller et al., 2022, provide a recent discussion).²³ Despite methodological challenges and data limitations, there is a general agreement that while increasing levels of concentration, including mergers, can be associated with higher efficiencies (through, e.g., economies of scale), they could also result in anticompetitive behavior in the absence of these efficiencies and other offsetting factors (e.g., technological efficiencies or pro-competitive gains),²⁴ whether through unilateral or coordinated actions (effects) (Hovenkamp and Shapiro, 2018; Nocke and Whinston, 2022).

In this section, we present key aspects and findings from the up-to-date literature that examine the effects of concentration on conduct and performance across markets upstream, midstream, and downstream the agri-food chain. Special attention is paid to potential market power exertion

²³ To illustrate these problems, note that both prices and quantities (market shares) depend on demand, supply, and the underlying factors that drive them. Economic theory indicates that these variables are jointly determined in equilibrium, which implies that a regression analysis of, for instance, price on market concentration, does not show the sort of causal effect that a researcher would aim; the regression estimate would pick up various possible correlations that occur due to variations in underlying demand and supply factors (e.g., income shocks, cost shocks). This non-causality is a different and more fundamental problem than endogeneity or simultaneity. In the problem of endogeneity, a clear causal path between prices and concentration can be established by a particular economic model but the causal factor (in this example, concentration) can also be correlated with unobserved factors not accounted for in the regression analysis. The problem of simultaneity occurs, instead, when the causal variable is simultaneously determined by the variable on the left side of the regression (in this example, price), as in the classic economic model of supply and demand. Practitioners have developed many possible solutions to both problems of endogeneity and simultaneity, which in different situations might include controlling for additional factors in a regression or using what are called instrumental variables. However, these solutions do not solve the sort of problem presented by a regression where there is no causal relationship to be estimated (O'Brien, 2017; Miller et al., 2022).

²⁴ Kolasky and Dick (2003) provides a historical progression of how efficiencies are considered in merger analysis in the US, recognizing that they should play a central role in merger review.

reflected in traditional conduct outcomes such as price or quality;²⁵ in coordination strategies among powerful actors (or tacit collusion); in the prevalence of unfair trading practices (UTPs), which is an issue of increasing concern especially in Europe;²⁶ and in performance indicators such as price margins (i.e., markups or differences between sales and purchase price) and profits.

3.1 Upstream

Fertilizers

Kim et al. (2002) analyze trade-offs between market power and cost efficiency in the US nitrogen fertilizer industry, using data from 1976 to 2000. The industry is characterized as a price-leadership model for oligopolistic competition (i.e., where there is a dominant firm among the few firms in the market) and the analysis shows that market power costs outweigh (by 55%) the benefits in terms of manufacturing cost efficiency associated with economies of size from increased concentration. The authors note that strong market power effects have implications for the stability of both the nitrogen fertilizer supply and farmers' revenues, as production is lower and the price paid by farmers is higher than if the market operated under perfectly competitive conditions. In their view, long-term contracts by fertilizer producers do not necessarily lead to a stable supply at a relatively low price; as natural gas price rises, which is the most important production cost, nitrogen fertilizers, putting supply at risk.

Hernández and Torero (2013) examine the fertilizer industry at the global level and assess the relationship between market concentration and prices. The authors note that despite the fertilizer industry is already highly concentrated, in further concentrated markets prices are even higher, which could limit fertilizer access to many smallholder farmers. Using annual data on urea prices and market shares for a panel of 38 countries for 1970-2002, the authors perform some simulations to show the potential positive effects of increasing global and regional competition in the industry

²⁵ In the case of a seller firm, market power refers, for example, to the capacity of the firm to set a price above marginal cost, while in the case of a buyer firm it refers to the capacity to pay the supplier a price below the marginal product value (i.e., behaviors that deviate from perfect competition conditions).

²⁶ An EU directive was adopted in April 2019 to regulate unfair trading practices in business-to-business relationships in the agricultural and food supply chain (EU Directive 2019/633).

on agriculture in low-income countries, which are highly dependent on imported fertilizer. For instance, a 10% increase in competition could increase fertilizer use by 13% to 19% and rural incomes by 1% to 2% in regions such as sub-Saharan Africa. Considering that local fertilizer distribution channels in developing countries are similarly characterized by a small number of traders, Hernández and Torero (2018) supplement earlier results discussing partial correlations between market concentration and price margins in Kenya. Using data collected in 2016 through interviews with wholesalers and retailers in major market centers across Kenya's Central and Western regions, the authors find a positive correlation between local market concentration and wholesale and retail price margins for different fertilizer products (CAN, DAP, NPK, and urea).²⁷ As more data become available, the authors encourage a closer look at the fertilizer industry and the relationship between concentration and prices in order to obtain a better understanding of competitive behavior along global, regional, and local supply and distribution channels that can help prevent potential market power exertion.

Likewise, the global fertilizer market has been showing unique record high prices over past major events. During the food price crisis of 2007-2008, the price spikes in the fertilizer industry were much higher compared to commodity prices in energy and agricultural markets (Hernández and Torero, 2013). In 2021-2022, fertilizer prices have spiked again and more than tripled in real terms since the start of the pandemic (while grains less than doubled) due to surging input costs, supply disruptions, export restrictions and sanctions, and the Russian invasion of Ukraine. This has upped concerns about fertilizer availability and accessibility, which heavily depend on a few major producers worldwide (see above), and, hence, the impact fertilizer shortage may have on global food security (Glauber and Laborde, 2022; Hebebrand and Laborde, 2022). Baffes and Koh (2022) expect fertilizer prices to remain high for some time to come. These exceptional price hikes, combined with past industry reports from Green Markets indicating record profits by leading fertilizer producers (with joint total revenues of over US\$50 billion per year),²⁸ invite for a more in-depth assessment of fertilizer firm behavior.

²⁷ CAN stands for calcium ammonium nitrate, DAP for diammonium phosphate, and NPK for complex fertilizer (nitrogen, phosphorous, potassium).

²⁸ Green Markets, fertilizer market data, <u>www.fertilizerpricing.com</u>.

Seeds

A study from the OECD (2018a) provides a comprehensive assessment of the effects of market concentration on prices and innovation in the seed industry based on disaggregated cross-country data for various crops, including cotton, maize, potato, rapeseed, soybeans, sugar beet, sunflower, and wheat and barley.²⁹ The statistical analysis using various specifications do not find clear evidence of a positive association between market concentration and seed prices, after controlling for crop-specific and country-specific effects on prices. According to this study, apparent structural differences in price levels between countries could be explained by quality differences and the presence of public plant breeders in some markets, besides varying market concentration levels. In the case of innovation, while there is a positive association between market size and innovation in plant breeding (number of new market entries),³⁰ the role of market concentration is less clear: the partial correlations are all imprecisely estimated and vary across specifications.

The study similarly reports a considerable degree of multimarket contact (i.e., overlap between the same group of players in different markets), which is worth paying special attention;³¹ a high degree of overlap between the same firms across different markets could increase the likelihood of (tacit) collusion between firms as well as reduce the contestability of markets by diminishing the likelihood of entry by outsider firms (Bernheim and Whinston, 1990; Fernandez and Marin, 1998).³² The data and analysis presented by OECD (2018a) covers mostly developed countries and some emerging economies, so for most developing and least developed countries, information

²⁹ Data availability for prices comprises several EU Member States and a selection of countries from North and South America (United States, Mexico, Brazil, Uruguay, Paraguay, Argentina), Oceania-Asia (Australia, Philippines, Thailand, Indonesia) and Africa (South Africa). In innovation, the analysis is focused on the EU.

³⁰ Marketing of seeds in the EU requires the variety to be registered in the national list of one Member State. ³¹ The degree of multimarket contact seems to be higher, for example, for maize, rapeseed, and sunflower seed but lower for wheat and barley. Unfortunately, limitations on their dataset made it difficult to assess the impact of multimarket contact on seed prices, as the dataset does not have full coverage across crops and countries. The authors further note that any empirical measure of whether firms have contact across different markets would tend to understate the true level of multimarket contact.

³² The study also remarks that in the context of the seed industry, multimarket contact can occur across several dimensions: firms in the seed industry can compete in agrochemicals (or even in non-agricultural products); firms competing in one geography may also face each other as competitors in other geographies; firms competing in the seed market for one crop may also compete in the seed market for other crops.

regarding market concentration or the potential impact of recent mergers in the seed industry on market outcomes is scarce.

Bryant et al. (2016) perform model-based simulations of mergers in the seed industries for maize, soybeans, and cotton in the US. While their method has some limitations (as discussed in OECD, 2018a),³³ the findings suggests that price increases tend to be greater in markets where the combined market share of the merging firms is higher, which points to a positive correlation between concentration and prices. The results further indicate that firms with smaller initial market shares are more likely to raise prices the most after merging.

Çakır and Nolan (2015) developed a theoretical model with both an input and complementary input sector upstream and an output sector downstream that can help to better understand the implications for producer and consumer welfare of market power exertion upstream the agri-food chain.³⁴ The authors compare market power exerted in the complementary input versus downstream sector and find that market power exercised by the supplier of the complementary input generates greater negative welfare effects than the same level of market power exercised by downstream firms. The intuition for the result comes from the differences between the information available to downstream firms and suppliers of complementary inputs or complementors. As downstream firms can exert oligopolistic power by using their knowledge of demand for the primary input, complementors can exert oligopolistic power by using their knowledge of derived demand, which comprises information on both demand for the composite commodity and supply for the primary input. Compared to the welfare distribution under perfect competition, the oligopolistic power stream series greater welfare losses to consumers and producers than welfare losses stemming from an equivalent degree of oligopoly or oligopsony

³³ Some caveats in the analysis include the high level of aggregation (the US as a single market) and assumptions that only the merging firms will increase their prices or that other factors (e.g., product choice or innovation) remain unchanged.

³⁴ The model focuses on a primary input sector, which consists of numerous suppliers (upstream firms), and allows for exertion of market power in a concentrated complementary input sector (complementors) as well as in the downstream sector (downstream firms producing a composite commodity using the two inputs). The model setting is an oligopoly/oligopsony for a homogeneous good in an imperfectly competitive market assuming Cournot (quantity) competition with symmetric firms.
market power from downstream firms.³⁵ Their results are discussed in the context of the Canadian grain-handling and transportation system, showing that benefits from regulation in that sector could potentially accrue to downstream firms more than to upstream suppliers for whom the regulation intended to achieve competitive outcomes.

3.2 Midstream

McCorriston (2014) summarized the literature in the context of the 2013 OECD Roundtable on Competition Issues in the Food Chain Industry (OECD, 2014), and concluded that "[e]mpirical evidence on the existence of buyer power is generally lacking."

A more recent compilation by Perekhozhuk et al. (2017) includes 38 studies on buyer and/or seller power in agricultural product markets, and only find modest deviations from competition in most of their conduct parameter estimates; the relatively low magnitude of supply elasticity estimates (i.e., implying that a demand reduction has a large impact on farm prices) explains why several studies tend to find large price effects, but only modest levels of market power.³⁶ The reviews by Sheldon (2017, 2018) also conclude that there is little robust empirical evidence of food processing firms exerting buyer power. Similarly, Sexton and Xia (2018) point out that empirical studies "have tended to find only mild departures from perfect competition". Nonetheless, they argue that today's market characteristics are not consistent with standard economic model assumptions and "likely cause market power to be less than would be predicted based on the highly concentrated structures of many modern agricultural and food markets." In a review of competition issues in agri-food value chains, Swinnen (2020) notes that there is "mixed" evidence of a systematic problem of buyer power at the expense of farmers. Deconinck (2021) reviews 42 relevant empirical studies regarding buyer power affecting farmers (16 of them not considered in earlier works by

³⁵ Or equivalently, for the same degree of market power, the welfare consequences from complementors' oligopoly power are the same as those due to the downstream firms' combined oligopoly and oligopsony market power.

³⁶ In the studies reviewed by Perekhozhuk et al. (2017), buyer power is exerted by restricting purchases to farmers in order to lower prices. The estimated price effect is thus a combination of the overall level of demand reduction and the impact the reduced demand has on farm prices (captured by the elasticity of farm supply). If farm supply is highly inelastic (i.e., the supply is not much affected by the price at which farmers are able to sell their product), even a small demand reduction could result in large price effects, as appears to be the case in the studies reviewed.

Perekhozhuk et al. (2017), Sheldon (2017, 2018), and Sexton and Xia (2018)) and only finds evidence of economically significant levels of market power and/or price distortions in some of them, as in the Greece food supply chain (Kaditi, 2012).³⁷

As Perekhozhuk et al. (2017) and Sexton and Xia (2018) highlight, the different methods to estimate price markups are based on different assumptions such that the resulting estimates can also differ significantly. Koppenberg and Hirsch (2021) note that if these assumptions are potentially inaccurate, estimates for market power and the resulting predicted welfare effects can be biased and recommendations for policy makers may be inaccurate. Sexton and Xia (2018) consider that the most promising methods to obtain reliable market power estimates avoid either estimation of conjectural elasticities and market conduct parameters or avoid estimating cost parameters.

There are also several studies estimating (asymmetric) price transmission across the chain and its relationship with market power.³⁸ In an edited volume on price transmission in agri-food markets, McCorriston (2015) concludes that the evidence regarding asymmetric price transmission is mixed and varies between sectors and periods, and there is no strong evidence of its association with buyer concentration. For example, in an empirical analysis of the Germany dairy sector, Holm et al. (2012) find substantial asymmetric price transmission, but the authors do not find a correlation with the strength of a brand (a proxy for market power), while the impact on profits is limited. A similar mixed conclusion is reached in the price transmission studies reviewed by Swinnen and Vandeplas (2015), but their conceptual framework is additionally helpful to understand that an important shortcoming in the price transmission literature is the assumption that factor markets

³⁷ Kaditi (2012) finds that buyer power is exerted to some extent by all firms in the supply chain in Greece, including the (relatively large) agricultural firms in her sample based on data from nearly 3,000 companies between 1998 and 2007. Her estimates show that food processing firms reduce prices by around 14%, while the corresponding wholesalers' and retailers' estimates are 27% and 42%, respectively.

³⁸ Broadly, this literature tests if price transmission depends on whether prices go up or down, and the hypothesis is that concentrated businesses in segments of the value chain can take advantage of differential effects (i.e., asymmetric pass through of prices) to extract rent. Deconinck (2021) discusses that at least when it comes to deriving conclusions on market structure, a limitation of this literature is the lack of a price transmission and market power link in theory. The review of price transmission research in the food chain (both empirical and theoretical) performed by Lloyd (2017) concluded that "[t]heoretical analysis highlights the fact that, without careful consideration of the range of factors involved, little can be inferred about market power from the low (or high) value of the price transmission coefficient alone".

work well and that contracts are enforced, and vertical coordination and search and monitoring costs are often ignored. The authors argue that these factors are widespread and are not only present in less developed regions, but also in many sectors in modern agri-food value chains where search costs can be significant. In the same vein, the authors stress that vertical constraints and supply chains that require contract-specific and critical investments by buyers rule out the traditional assumption that weaker price transmission is associated with lower supplier welfare, such that the belief that powerful intermediaries in the supply chain are capturing rents is no longer applicable.

As pointed out by Deconinck (2021) and more recently by Crespi and MacDonald (2022), the US meatpacking industry has received special attention over the past decade with an important number of studies examining the effects of concentration on market outcomes, including studies commissioned by the government, due to persistent anti-competitive allegations concerning important players in the industry.³⁹ Yet, a review of this literature concludes that concentration in procurement of livestock (cattle or hogs) has not adversely affected prices received by producers or prices paid by consumers. There is evidence that producers may be actually better off because of lower processing costs due to concentration and introduction of technical innovations (Wohlgenant, 2013). A more recent work by Ji et al. (2017) is consistent with these findings, and further suggests that markets have become more competitive since the 1980s (despite higher levels of concentration). In the same industry in Canada, an empirical analysis by Rude et al. (2011) shows that market conditions did not deviate from competitive conditions between 1992 and 2001, in line with earlier studies which similarly did not find evidence of market power as long as

³⁹ Over the past months, the US meatpacking market has regained political attention. On July 2021, President Biden issued Executive Order (EO) 14036, "Promoting Competition in the American Economy", and in the ensuing months, federal agencies have begun to implement several initiatives to address competition issues across the economy, including a more competitive meat and poultry supply chain. On January 2022, the US Department of Justice and US Department of Agriculture issued "Shared Principles and Commitments to Protect Against Unfair and Anticompetitive Practices", including a statement that commits to jointly develop a centralized, accessible process for farmers, ranchers, and other producers and growers to submit complaints about potential violations of the antitrust laws and the Packers and Stockyards Act; and to vigorously enforce the laws that protects them from unfair, deceptive, discriminatory, and anticompetitive practices (available at: https://www.justice.gov/opa/pr/justice-department-and-agriculturedepartment-issue-shared-principles-and-commitments-protect). On March 2022, the US Department of Agriculture announced a supplemental funding to promote competition and create more markets for local and regional food producers by expanding and strengthening opportunities to sell to institutions, e.g., hospitals (available at: https://www.usda.gov/media/press-releases/2022/03/01/usdauniversities. announces-supplemental-american-rescue-plan-funding-available).

Canadian producers had the option of shipping live cattle to the United States; however, they did find evidence of buyer power during the temporary ban on Canadian live cattle imports into the US in May 2003 (after a case of bovine spongiform encephalopathy in Canada).

In a wide-ranging study that covers several elements relevant for the analysis of agri-food value chains in low- and middle-income countries, Barrett et al. (2021) point out that most studies on market power in agri-food value chains have focused on developed economies. The study includes a review of the empirical literature for developing countries' agri-food distribution markets, and remark that the degree to which perfect competition or market power prevails has been one of the longstanding preoccupations. Within very localized areas, competition typically prevails, although remoteness, financial liquidity constraints, and associated credibility and reputation issues can confer considerable market power in niches that require significant capital or are characterized by non-trivial economies of scale or scope, especially long-haul, large-scale trading (Dillon and Dambro, 2017; Casaburi and Macchiavello, 2019; Bergquist and Dinerstein, 2020; Casaburi and Reed, 2017, 2022). Aspects like high search, storage and contracting costs can sharply limit competition over longer distances and across seasons, limiting spatial and inter-seasonal competition (Barrett, 1997; Moser et al., 2009). In these contexts, intermediaries can take large surpluses in detriment of consumers and farmers. Barrett et al. (2021) emphasize that across different countries and periods, food marketing intermediaries are typically distrusted as predatory agents rather than celebrated for facilitating welfare-enhancing transactions, while the low quality of infrastructure and long distances in many markets can indeed foster market power.⁴⁰

Another important dimension of potential market power abuse studied in the agri-food industry are UTPs and their impact on farmers, although the evidence is mainly anecdotal and, in most cases, outdated (Falkowski, 2017). UTPs are various types of practices that can be linked with abuse of market power and can be classified into four categories: i) the retroactive misuse of unspecified, ambiguous, or incomplete contract terms; (ii) the excessive and unpredictable transfer of costs or risks of a trading partner to its counterparty; (iii) the misuse of confidential information; and (iv) the unfair termination or disruption of a commercial relationship (Swinnen, 2020). In most surveys to farmers reviewed by Di Marcantonio et al. (2018), the majority of respondents have

⁴⁰ Intermediaries, however, may also bear large operating costs.

experienced one or more UTP, but as Sexton (2017) remarks the surveys generally collect the view of one party of the transaction, which difficult assessing the problem from a more comprehensive perspective. Deconinck (2021) adds to this problem that the definition of UTP can vary across countries, and not all UTPs are equally common or severe. For example, nearly all (98%) dairy farmers from France, Germany, Poland and Spain surveyed in Di Marcantonio et al. (2018) reported at least one UTP;⁴¹ by comparison, the share of respondents reporting experiencing UTPs during or after contract execution is considerably lower.⁴² To further inquire in this finding, estimations using the aforementioned survey of dairy farmers, presented in the work by Di Marcantonio et al. (2020), show that contract completeness increases the likelihood of farmers reporting that their contracts with processors include practices that may be considered as UTPs and, in contrast, contract completeness does not seem to affect UTPs during the contract execution or its termination. According to the authors, these results support the hypothesis that surplus extraction is generally done through the inclusion of specific contract terms, while given the signed contract, UTPs are much less likely during contract execution and its finalization.

Deconinck (2021) also notes that in surveys regarding the prevalence of UTPs, the rates obtained in EU countries are relatively high. In addition, the author remarks that several in-depth studies of competition authorities have found instances of UTPs in agri-food sectors across numerous European countries (Bulgaria, the Czech Republic, Germany, Latvia, Poland, Slovenia, and Sweden), as well as in the wine grape sector in Australia. The study discusses a recent proposal to improve knowledge on UTPs, based in Russo (2020), that combines a broad-scope empirical analysis across sectors within each country (e.g., with a general survey) and more in-depth studies for specific sectors (e.g., using expert panels to design sector-specific surveys), including interviews and economic analysis to achieve a more efficient monitoring system for UTPs. More information can further help to better understand the actual impact of UTPs on consumers and

⁴¹ Specifically, almost 90% of farmers reported that the contract does not define any safeguard for the farmer if the buyer fails to fulfil the contract, while close to 20% of farmers reported that the price is set unilaterally by the buyer in the contract. Deconinck (2021), however, remarks that having an incomplete contract could not be considered itself an UTP, and the same applies for the price being unilaterally set by the buyer (Sexton, 2017).

⁴² For example, only 2.7% of farmers reported that the buyer unilaterally changed the price; 1.2% reported a unilateral change in required quality; and 1.5% reported a unilateral change in required quantity. These figures are remarkably low in a sector where UTPs were thought to be common.

suppliers and disentangle if some practices described as unfair, may in fact have an ambiguous economic effect. For instance, no consensus exists in the literature regarding how efficiency and income distribution are affected by certain forms of market power conduct, such as "reverse margin" practices (Falkowski, 2017).⁴³

Swinnen (2020) similarly discusses the evidence regarding the effects that requirements of standards can have on welfare, as a way dominant firms could gain additional market power. The author exemplifies this situation noting that suppliers are often confronted with a limited number of buyers willing (or able) to reward high standards because these (and the associated quality premiums) are, for example, company-specific (Henson, 2006; Smith, 2009). Thus, effects on efficiency attributed to solve asymmetric information problems can be offset by market power gains and implementation costs of the standard or label (Bonroy and Constantatos, 2015), such that conclusions regarding aggregate effects are mixed. Swinnen (2020) add that most studies do not focus on the distribution of welfare between producers and consumers but rather on total welfare. Only a few explore the effect of labels on the distribution of surplus, such as Bonnet and Bouamra-Mechemache (2016) that consider the effect of an organic label on the share of surplus created along the value chain between manufacturers and retailers with an explicit discussion of bargaining power.

Certain modern commercial relationships (e.g., contract farming or participation in supermarket channels) tend to enhance market concentration and vertical integration of the agri-food chain. Research of such relationships in developing countries has led to recommendations that public policy should be supporting the establishment and maintenance of this type of contractual arrangements, especially when involving small-scale farmers. Such recommendations stem from the evidence provided by those studies that inclusion and participation in contractual arrangements, even in regions with different agro-ecological conditions, crops, characteristics of buyers' firms and contracting practices, tends to (i) increase households income (Reardon et al., 2009; Miyata et

⁴³ "Reverse margin" practices are contractual clauses which require a supplier to pay a fee to the buyer, for example, in exchange for services offered by the buyer such as promotional activities. Deconinck (2021) recall that both theoretically and empirically it is less clear whether such practices negatively affect efficiency. If a stronger party can negotiate a more advantageous price, it is unclear why it would resort to additional contract terms to extract more surplus if doing so has a negative effect on efficiency.

al., 2009; Bellemare, 2012), (ii) improves productivity and efficiency (Rao et al., 2009), and/or (iii) increases the portfolio of suppliers' assets (Michelson, 2013). This stream of research also emphasizes the importance of providing technical advice (extension services) and access to inputs (e.g., seeds, agrochemicals), and contextual elements such as the availability of infrastructure, proximity to markets, involvement of NGOs or donor-funded projects in the development of supply chains, or the fact that in developing countries the retail sector is still developing (Michelson, 2013).

Yet, while the spread of supermarkets and retail companies generally have a positive impact on participating smallholder farmers in developing regions, Barrett et al. (2012) note that only a small fraction of smallholders manage to access modern commercialization and dynamic markets while many other remain in the semi-subsistence system. Ebata and Hernández (2017) stress that commercialization by small-scale agricultural producers is often hindered by high transaction costs related to coordination failures between agents (whether private and/or public) and to regulatory requirements. In addition, many small-scale producers in developing countries lack appropriate managerial capacity and lack access to adequate production technology and infrastructure, preventing them from effectively accessing markets and commercialization.

In the case of Sub-Saharan Africa, Bjornlund et al. (2020) further point the importance of improving governance structures and market integration for a transition from low-producing, unprofitable agriculture to modern, efficient, and profitable systems. The authors similarly recognize the role of history in preventing the development of agro-processing and manufacturing industries (with African colonies having been used to source raw products for export to colonial powers where processing would occur thus favoring concentration of manufacturing and processing activities among a smaller number of countries) as well as geopolitical, financial, and trade issues that constrain African countries from introducing policies to adopt modern technologies and increase agricultural productivity.

Hence, conclusions drawn from participation in modern channels are typically presented with caution, and often recommend policies that include improved credit, extension services, and input delivery systems, which can help farmers to sustainably upgrade their production technology.

However, more research is necessary to better understand the implications of market concentration in contexts of increased market linkage and commercialization activities with the retail (supermarket) sector. As remarked by Swinnen (2020), with search and contracting problems, increased competition may complicate coordination and contracting since hold-up opportunities and/or transaction costs increase, which can apply both to developed and developing market conditions.⁴⁴ On this point, Macchiavello and Morjaria (2021), recently analyzed relational contracts between upstream farmers and downstream mills in the Rwanda coffee chain and show that additional competition by mills increases the likelihood of farmers defaulting on contracts, makes farmers worse off, and results in lower aggregate quantity of coffee supplied to mills.⁴⁵

Lastly, the role of cooperatives in generating or countervailing market power has received some attention in the literature. Lee and Van Cayseele (2022), for instance, recently estimate firm-level markups and markup volatility to identify the market power of farmer and processor cooperatives in the Italian fruits and vegetable and dairy sectors. The period of analysis (2007-2014) is characterized by major price volatility in global agricultural and food markets that also allows to assess whether cooperatives could function as a management strategy against risk. The estimation results show a tradeoff in cooperatives' roles between obtaining market power and stability, i.e., cooperatives obtain stability in their markups at the expense of lower markup levels. Farmer cooperatives seem to place a higher importance on profit stability due to the inherent uncertainties that already exist in agricultural production. Processor cooperatives, instead, obtain more market power in the fruits and vegetable sector that appear to arise from the product differentiation strategy of these cooperatives, while processor cooperatives in the dairy sector obtain higher stability but their market power is lower.

⁴⁴ Mérel and Sexton (2017) observe that the length of processors' commitments (in the form of contracts) is shorter than the lifespan of the specialized sunk assets required by farmers to meet the processors' needs. This leads to the so-called holdup problem: anticipating that processors will exploit short-run input supply inelasticities (exacerbated by perishability and limited selling outlets), farmers underinvest in specialized resources relative to the social optimum.

⁴⁵ The authors, however, remark that the benefits of competition might be hampered by the presence of other market failures (e.g., imperfect input and financial markets) and point to the possibility of excessive entry when contracts are hard to enforce. In this case, a mill's efficiency requires well-functioning relationships with farmers in which the sale of coffee cherries at harvest is bundled together with the exchange of services before, during, and after harvest in both directions.

3.3 Downstream

Koppenberg and Hirsch (2021) study mark-up pricing behavior in the food retailing sector in the EU, specifically in Finland, France, Italy, Portugal, and Sweden, for the period between 2010 and 2018. The authors use two different mark-up identification methods (the stochastic frontier approach-SFA and the production function approach-PFA).⁴⁶ These countries were chosen based on data availability and the fact that their retail sectors exhibit different degrees of concentration. Their analysis indicates that the two methods lead to significantly different average mark-up levels, which provide differing assessments of the state of competition in the analyzed sectors. Accordingly, the authors encourage other researchers examining market power to discuss assumptions in greater depth and to be careful when submitting policy recommendations based on consumer welfare loss calculations stemming from markup estimates.

Bonanno et al. (2018) reviews the predictive capability of traditional models and tools to assess and measure market power in the agri-food sector and remarks the importance of thinking in terms of other paradigms, especially bargaining power rather than (or in addition to) market power. Similarly, in their review of pricing strategies among food retailers in developed countries, Hamilton et al. (2020) assess theoretical and empirical evidence for retail pricing and the vertical relationships that have emerged among retailers, food manufacturers, and farmers.⁴⁷ The study presents several examples of strategies to price above marginal cost and considers an emerging body of research that frames vertical relationships between suppliers and retailers in terms of bargaining power (for example, using the Nash bargaining model) instead of the traditional market power approach.⁴⁸ The authors remark that future research should consider new ways of thinking

⁴⁶ The authors chose the EU food retailing sector for the analysis since antitrust authorities have recurrently initiated investigations against food retailing companies in several member states, for example, the Czech Republic and Finland (OECD, 2014) or France (European Commission, 2019).

⁴⁷ The analysis of consumer behavior in multiproduct retail markets includes consumer search, habit formation, and reference pricing. The authors also discuss retail market outcomes for price discrimination, price fairness, and price obfuscation; and then turn to examine the relationships between retailers and food manufacturers through bargaining outcomes, market foreclosure, and slotting allowances, and discuss the resulting implications for retail-price pass-through.

⁴⁸ In the standard Nash bargaining model, the estimator of bargaining power is a function of agreement and disagreement payoffs, the latter being the payoffs earned by each player in the absence of an agreement. In satisfying the solution, this model assumes simultaneous and bilateral negotiations such that, in equilibrium, no party wants to renegotiate (Horn and Wolinsky, 1988). For instance, the solution of the Nash bargaining problem between retailers and manufacturers, assuming Bertrand rivalry in downstream retailers, yields a

about vertical relationships among farmers, distributors, manufacturers, and retailers, and more closely examine how market power is consolidated in an increasingly concentrated food system at each of the various points in the supply chain.

For example, in an application of the Nash bargaining model to assess vertical relationships between Chilean retailers and coffee manufacturers, Noton and Elberg (2013) challenge the view that large supermarket chains can extract most of the surplus generated across the value chain. During their period of analysis, the two retail chains included in the study captured more than 60% of the total supermarket sales in the country and 88% of the coffee sales at supermarkets.⁴⁹ Yet, the authors find that small coffee manufacturers obtain a sizeable fraction of the surplus, consistent with a strong bargaining power and despite their small market shares. In addition, in a context of a volatile surplus due to large swings in international coffee prices, manufacturers absorb most cost shocks, which could be due to their relative advantage, compared to retailers, in terms of inventory or hedging management. In this line, the authors suggest that more research is necessary to examine the relationship between cost absorption and inventory and/or hedging behavior in vertical channels under surplus volatility (as in bread-wheat, poultry-corn, chocolate-cacao, cooking oil-oilseeds value chains), exploring the sources of bargaining power with richer datasets in terms of the number and characteristics of bargainers.

More generally, advancing in this area of research could complement with the relatively recent findings in the analysis of strategies of large firms, such as supermarkets in food retail. Hausmann and Leibtag (2007) show that retail food prices fall when a Walmart store opens in a specific geographic location using US household data for the period 1998 to 2001.⁵⁰ Based also on data

single parameter that describes the share of the total margin that is appropriated by either the manufacturer or the retailer, depending on the relative bargaining strength of each party. Substitution and complementarity (i.e., the structure of demand) are key factors determining the payoffs and can have significant effects on estimated bargaining-power parameters, as shown by Richards et al. (2018a) using data on multi-category soft drink purchases among households in France.

⁴⁹ One supermarket chain was acquired by US-based Walmart, and the other had started an internationalization process to several other Latin American countries, including Argentina, Brazil, Colombia, and Peru.

⁵⁰ The study uses a customized subset of the ACNielsen Homescan scanner panel data for randomly selected households across the US. The authors further argue that Walmart supercenters have had the biggest impact on food retailing as they compete most closely with traditional supermarkets and offer many identical food items at an average price about 15%–25% lower than traditional supermarkets.

from US markets from late 1980s to late 1990s, Jia (2008) develop an empirical model showing that expansion of a large retailer could make unprofitable or unable to recover sunk costs for roughly 50% of discount stores in the same markets. These findings have encountered explanation in Smith and Thomassen (2012) and Rhodes (2015), who argue that product complementarity influences pricing strategies among retailers downstream. Internalizing cross-product pricing effects on the intra-retailer margin with product complementarity leads to lower retail prices as retailers have an incentive to drive volume rather than margins. As this food business works with small margins, reaching high volume of sales is essential; this similarly provides support to why supermarkets may enter into price wars to wipe out competitors in local markets or, alternatively, profile products for niche markets (e.g., gourmet foods; organics) or store "feeling" (including convenience of selling assortments as broad as possible of food and non-food items).⁵¹ On the other hand, regarding inter-retailer strategies, Richards et al. (2018b) present evidence in the other direction showing that product complementarity is associated with anti-competitive effects and is a source of market power for retailers; the authors point out that previous evidence on single category purchases does not take into account that once customers have made the trip to the store, product complementarity could lead to softening competition and higher equilibrium prices relative to the case where customers purchase products with independent demands. Yet, Richards et al. (2018a) argue that none of these studies focused on vertical relationships between multiproduct retailers and manufacturers. To incorporate this dimension, the authors develop an empirical model of bargaining that includes products from four different sub-categories in the softdrink industry in France and show that product complementarity can lead to a higher retailer share of the total margin. This means that retailers can enhance their bargaining power, relative to manufacturers, by selling complementary products across categories. Retailers understand that manufacturers may need to offer a broad array of products across different categories to extend their company (brand) appeal;⁵² hence, knowing the pressures faced by manufacturers, retailers can negotiate accordingly and are able to extract greater rents in the vertical channel. As brands expand across related categories, and even related channels, retailers can take advantage that manufacturers need to be omni-present to stay in consumers' minds. Manufacturers negotiate with

⁵¹ The emergence of larger retail-store formats enhances the opportunity for retailers to generate rents from cross-category purchases through price discrimination (Hamilton et al., 2020).

⁵² For instance, food processors may brand (or set some standard) for their own products and compete for market shares through such branding (alleged quality) in a way to enjoy higher margins for such products.

retailers' incentives in mind, so complementarity between products should imply more retailer bargaining power and lower manufacturer power, relative to the usual substitute-products situation.⁵³

In the aforementioned research review of Sheldon (2017), there is evidence suggesting that US food retailing may already be undergoing further structural change in recent years, with traditional food retailers finding difficulties to follow a strategy of product variety expansion, given competitive pressure on their margins from big retailers and the increasing incentive to focus on a narrower range of high-quality fresh unpackaged foods (Watson, 2016), combined with the fact that online food retailing is expected to grow more rapidly over the next decade compared with traditional food retailing. On this matter, Hamilton et al. (2020) point out that retailers such as Walmart offer a wide range of in-store services, potentially increasing their opportunities to raise overall store margins through other mechanisms, like price obfuscation strategies (Piccione and Spiegler, 2012); the authors further remark that pricing complementarity strategies not only provide an alternative explanation for the rise of large retailers, but also help to explain increased investment in the food retailing industry (e.g., the entrance of Aldi into the US market and the purchase of Whole Foods by Amazon).⁵⁴

In a broader assessment of macroeconomic implications of market power in the US, De Loecker et al. (2020) document the evolution of market power across all sectors of the economy using firmlevel data for publicly traded firms over the period 1955-2016 and data from the US Census on manufacturing, wholesale, and retail.⁵⁵ The authors show that markups started to rise after the

⁵³ When products sold by one manufacturer are complements downstream, the disagreement profit, which is the amount earned if the parties fail to agree, is lower with complementarity than under strict substitutability. Lower disagreement profit implies a higher opportunity cost of agreeing. As a result, manufacturers are essentially keener to arrive at a negotiated solution, such that their bargaining power is lower (Richards et al. (2018a).

⁵⁴ Hamilton et al. (2020) also point out that the ability to search more efficiently for desired product attributes in online markets can sharpen inter-retailer product differentiation and reduce the elasticity of demand for foods sold online, thereby raising retail margins.

⁵⁵ The study is based on Compustat firm-level financial variables of all US publicly listed companies active at any point during 1950-2016. Census data for manufacturing (NAICS codes 31-32-33), wholesale (NAICS code 42), and retail (NAICS codes 44-45) is used to perform robustness checks and verify the extent of selection bias in the sample of publicly traded firms. The authors estimate that these three sectors account for roughly 20% of the US Gross Domestic Product (GDP). The figures for manufacturing and wholesale using Census data are in line with the main results using data from publicly traded firms; but in

1980s, from 21% above marginal cost to 61% in 2016, and the increase is mainly driven by sharp increases in the upper tail of the markup distribution (i.e., in the upper percentiles). They also find a reallocation of market shares from low- to high-markup firms within industries as well as an increase in the average profit rate from 1% to 8%. These estimates are based on a method developed in De Loecker and Warzynski (2012) that uses information from firms' financial statements and does not require any assumptions on demand and how firms compete.⁵⁶ As data requirements become more accessible, this method could be more suitable for analyzing specific industries across the agri-food sector in other countries, compared to more standard approaches using structural methods.

In developing countries, Atkin and Donaldson (2015) generate estimates of the relative scale of intra-national trade frictions and show their implications for market power. Using data from Ethiopia, Nigeria, and the US, the study analyzes barcode-level price data for many processed food items (that permit to control for unobservable product quality differences) and identify the exact point of origin of each product matched with observed consumer retail prices in spatially varied markets (to account for the distance each product travelled to where its price was observed). The authors find that the effect of distance on trade costs is 4-5 times larger in Africa than in the US, which permits to infer the extent of local market power exertion. On this matter, Barrett et al. (2021) indicate that these local trade frictions lead to intermediary market power such that traders capture most of the gains from reduced costs of imports that result from lower international trade barriers.

Finally, as part of the nascent experimental literature to learn about market competition, Busso and Galiani (2019) examine the impact of randomly induced entry of small retail shops on prices and quality in the Dominican Republic. By experimentally varying the number of new stores authorized to accept debit card payments of a conditional cash transfer program, the study shows that incumbent stores reduced prices from 2% to 6% and exhibit a statistically significant

the retail sector, the pattern of markups obtained is flat or slightly increasing until 2002, and for 2012 show a sharp increase that is not observed for publicly traded firms.

⁵⁶ In this method (PFA), markups are obtained by exploiting cost minimization of a variable input of production, which requires defining the production function (i.e., technology) to obtain the output elasticity of (at least) this variable input of production.

improvement in customer-reported service quality, with greater effects in places with more new entrants. The authors argue that competition seems to have driven part of the customers away from incumbent retailers.

Overall, most reviews of market power in agri-food value chains have mainly focused on midstream and downstream markets in developed economies, where there is more data availability, and do not find strong evidence of market power exertion, especially buyer power from food processing firms. Few studies find evidence of economically significant levels of market power and/or price distortions. The evidence regarding asymmetric price transmission and its relationship with buyer concentration is similarly mixed and varies across sectors and periods. In developing countries, the literature is scarcer but shows the importance of contextual elements such as remoteness, financial liquidity constraints, credibility, reputation issues, and high transaction costs in conferring considerable market power, where intermediaries can take larger rents from consumers and farmers. Similarly, the emerging experimental evidence in developing countries on market competition and performance is mixed although it is not generalizable as it has largely focused on establishing the relevance of a specific market mechanism or characteristic in the value chain under study; e.g., the aforementioned work by Busso and Galiani (2019) showing that randomized entry in grocery stores in Dominican Republic reduces prices and improves quality; randomized subsidies to cocoa traders in Sierra Leone revealing a relatively competitive intermediary (traders) segment, with a low degree of differentiation and a large market size, but showing substantially less competition downstream the value chain (e.g., wholesalers, exporters), as shown by Casaburi and Reed (2022); and randomized cost shocks and subsidies to maize traders in Kenya showing a high degree of market power and rent capture among intermediaries while exogenously induced firm entry has negligible effects on prices, as presented by Bergquist and Dinerstein (2020). See Reed (2022) for additional discussion on the possibilities that randomized studies and experimental design can offer to research questions related to market competition, welfare, and policy. More generally, and borrowing a term used by Macchiavello et al. (2022), the literature on food value chain transformation have not sufficiently "cross-pollinated" to empirical industrial organization. The authors argue that applying these advanced methods to food value chains in developing countries is now relevant because of the transformation (including favoring concentration) that has occurred and possibly due to the increasing availability of data from surveys of farms, processors, and wholesalers, and scanner data from some retailers.

Upstream, high levels of concentration in fertilizer and seed markets, combined with elevated levels of multimarket contact in those industries (where big multinational firms compete with the same rivals in multiple markets) are a matter of concern in terms of the effects that reduced competition or coordinated actions could have on fertilizer uptake or, more generally, on agricultural productivity. Record high fertilizer prices over the past and recent major events, such as the Russian aggression against Ukraine, further rise these fears, including global food security concerns, in a context where supply disruptions and export restrictions or bans to one of the few major worldwide producers can provoke immediate impacts on global fertilizer availability and accessibility.

In the same vein, as argued by O'Connor, (2021), the global pandemic has spread preoccupations derived from disruptions and shortages in agri-food supply chains and the increasing presence of larger and powerful downstream buyers across the globe. While achieving efficiency is a desired objective from both an allocation of resources perspective and for obvious reasons regarding its capacity to achieve low prices and welfare enhancing features from the perspective of consumers, there is an increasing need to further understand the implications of concentration among a few powerful buyers in the agri-food chain, at the expense of farmers, other producers, and workers.

In addition, the relatively few companies purchasing bulk agricultural commodities for international trade (pointed in Section 2.3) merits some attention in a context where global wheat prices have been recently increasing and fluctuating substantially (FAS-USDA, 2022), after market reactions to announcements about the Indian wheat export ban and ongoing negotiations regarding Ukraine's ability to ship, while Russia is still expected to be the main wheat exporter in 2022-2023 (despite economic sanctions and export taxes). Murphy et al. (2012) note that the few major trading companies act as midstream buyers who purchase high volumes and have leverage in terms of setting prices, particularly with farmers with whom they contract directly, as well as with grain elevators to which farmers in industrialized production systems deliver their grain. The authors further remark the role of the four major corporations (ABCD firms) in the so called

"financialization" of commodity markets and on potential speculation and food price volatility, which is worth exploring.⁵⁷

As more disaggregated data can be accessed or compiled (including detailed transaction data), new approaches and methodologies are being developed to better understand the complex relationship between market concentration, conduct, and performance at various points in the agri-food chain among input suppliers, farmers, distributors, food manufacturers, and retailers. These include an emerging body of theoretical and empirical research that frames vertical relationships between suppliers and retailers in terms of bargaining power instead of the traditional market power approach; recent developments in markup estimation methods that discuss assumptions in greater depth and recommends being careful when submitting policy recommendations based on consumer welfare loss calculations; and novel approaches based on experimental designs that represent new streams of research on market concentration and power and their welfare implications, particularly in less developed countries where data are typically collected from the field (as data availability is very limited) in more controlled settings. More research is also necessary to assess the impact of UTPs along the agri-food chain, i.e., not only on farmers but on other suppliers down the chain and consumers. Efficiency and rent distributional implications of some practices are similarly still not clearly understood in the literature. Although beyond the scope this study, studies analyzing the impact of concentration on conduct and performance in the food service industry (e.g., restaurants, fast-food chains) could be another interesting avenue for future research as the literature on this topic is very scarce.

⁵⁷ See Cheng and Xiong (2014) for an extensive review of the impact of financialization on commodity prices. Robles et al. (2009) also examined the role of speculation on agricultural commodity prices during the food price crisis of 2007-2008 and show that speculative activities might have been influential.

4. Implications of market concentration for food security and nutrition and environmental sustainability

This section assesses the potential implications of market concentration on food security and nutrition and environmental sustainability. In the case of food security and nutrition, the analysis pays special attention to access to healthy diets and reviews the available evidence both among smallholder farmer households and urban consumers considering the changing trends in commercialization channels and the food retail sector, particularly with the rapid expansion of supermarkets that have increased vertical integration and coordination favoring concentration along the chain.⁵⁸ In the context of environmental sustainability, the analysis focuses on reviewing the existing literature on the implications of concentration and/or vertical integration or participation in modern commercialization channels across the agri-food chain on environmental issues, including food loss and waste (FLW).

4.1 Implications for food security and nutrition

As pointed out earlier, the modernization of commercialization channels over the past decades has promoted the establishment of supply agreements with small farmers, with the potential to improve their welfare (e.g., Reardon et al., 2009; Rao et al., 2012). Several studies have examined the impact of participation in supermarkets' value chains on income, consumption expenditure, labor market insertion, and agricultural productivity (e.g., Miyata et al., 2009, Neven et al., 2009; Rao and Qaim, 2011; Bellemare, 2012; Michelson, 2013; Rao and Qaim, 2013; Andersson, et al. 2015; Mossie et al. 2021). Most of these studies conclude that supermarkets can contribute to rural economic growth, increased incomes, and modernization of the small farm sector. However, the evidence is scant regarding changes in dietary habits or nutritional status among smallholder households associated with engagement in supermarket chains (Gomez and Rickets, 2013; Popkin, 2014; Chege et al., 2015).

⁵⁸ Following Lee et al. (2013), a healthy diet can be defined as a diet that provides recommended amounts of foods, nutrients, and other food components, within estimated energy requirements, to promote normal growth and development in children, reduce risk of obesity and noncommunicable diseases, and promote optimum well-being at all ages, consistent with national dietary guidelines/recommendations.

Most of the recent research in developing countries about the impact of small farmers participation in modern value chains on food access and consumption has focused on African and South Asia with mixed results. Carletto et al. (2017) use data from Tanzania, Uganda, and Malawi and find little evidence of an association between increased commercialization and improved nutrition among farm households based on children's anthropometric measures, household per capita food expenditure, and household per capita caloric consumption. In Tanzania, Herrmann et al. (2018) investigate the impact of market linkages on agricultural income and food security. Even though farmers who have access to markets exhibit higher crop yields and receive better prices, there are no significant differences in household food security measured through weekly total per capita food consumption, household dietary diversity, or a household food insecurity access scale. Mishra et al. (2018) assess the impact of contract farming on smallholders in India and, among other results, find that its adoption increases food security measured through a higher share of food expenditures on total expenditures. Debela et al. (2022), show that the impact of contract farming in palm oil on smallholder nutrition in Ghana depends on the type of contract employed. Resourceproviding contracts that include in-kind credits and technical assistance have larger effects than marketing contracts, which can even have negative effects in some cases.

Literature that specifically address the relationship between supermarket contracts and smallholder farmer's nutrition is more conclusive, but equally mainly focused on African countries. For example, Chege et al. (2015), compare nutritional indicators between farm households with and without supermarket contracts in Kenya and report that participation in supermarket channels can increase household income and diversify crop production, thereby improving household nutrition. Ochieng (2017), also using data from Kenya, analyzes the situation of vegetable producers in Kenya and finds that having contracts with supermarkets increases household income and diversity as well as calories consumed and zinc intake. Onyeneke, et al. (2020) examine the case of Nigerian poultry producers and find that the availability of modern food retail markets, including supermarkets as well as fast food firms, hotels, and convenience stores, correlates with better quality diets. Factors contributing to healthier diets include an increase in dietary diversity, a growth in poultry consumption and the production of vegetables, and changes in household gender roles.

In sum, the association between participation in modern commercialization channels, especially supermarket channels, and access to better diets among small farmers appears to be positive, but certainly the literature is still relatively limited to derive more generalized conclusions. The impacts may be context specific and could be affected by a variety of factors such as contract terms and farmers' conditions. More research is required in this area to explore these factors. In the same vein, it is important to extend the research to other countries or regions, as the effects may vary across areas considering the major role of geographic location and accessibility on effectively integrating small farmers into global supply chains. In addition, future work should attempt to capture more detailed data, including macro and micronutrient data, to assess impacts beyond consumption of specific food groups and dietary diversity.

On the other hand, the rapid growth of urban populations worldwide has contributed at the same time to a significant transformation of food retail systems to attend the increasing demand for a wide range of food in fast-growing cities, including many developing countries. Part of the transformation has been led by the wide expansion of supermarkets in urban and peri-urban areas (i.e., the "supermarket revolution"), which has changed the nature of distribution networks and supply chains while seeking to satisfy consumer food demand for year-round in a timely and efficient manner (Reardon et al., 2003).⁵⁹ Supermarkets have modern procurement systems, such as buying in bulk to lower costs and improve efficiency (i.e., offering competitive prices), and impose private standards to raise quality (Reardon and Timmer, 2007), but their diversification of food availability, including both healthy fresh foods rich in essential nutrients and less healthy energy-dense and nutrient-poor foods (usually processed), can also have important implications for the health and nutrition of the population, particularly the poor.

In this context, several studies have reviewed how changes in the food retailing system due to the increasing presence of supermarkets is correlated with variations in urban consumption and dietary

⁵⁹ While supermarkets started to proliferate in developed countries (such as the United States and Canada) since the mid of the twentieth century, their expansion in developed countries started in the 1990s and occurred in four "waves": a first wave in South America, East Asia (excluding China), and South Africa in the early 1990s; a second wave in Central America, Mexico, and Southeast Asia in the mid 1990s; a third wave in China, India, Vietnam, and parts of Africa in the late 1990s to early 2000s; and a fourth wave in other parts of Africa and South Asia in the mid to late 2000s (Minten and Reardon, 2008).

patterns. Hawkes (2008) performs a wide-ranging literature review and indicates that supermarkets can make a more diverse diet available and accessible to more people but can also promote the consumption of energy-dense and nutrient-poor, highly processed foods among poor populations; the author concludes that supermarkets overall encourage consumers to eat more, despite the type of food. Monteiro et al. (2013) shows that ultra-processed products (i.e., energy-dense, fatty, sugary or salty and generally obesogenic foods) are increasingly dominant in high- and middleincome countries mainly driven by transnational food manufacturing, retailing, and fast-food service corporations. Popkin (2014) indicates that the rapid growth of the retail sector in middleand low-income countries is contributing for processed and packaged food to rapidly reach food insecure and overweight populations, which can pose major challenges to the health of the poor. Similarly, Qaim (2017) points out that although there is still an important presence of wet markets and traditional grocery outlets in several developing countries, the growing role of supermarkets and processed foods can contribute to overweight and obesity among consumers. Reardon et al. (2021) argue that the increase in stunting, overweight, and obesity in Africa is linked to the increase in the consumption of ultra-processed foods, which is driven by the expansion of the food processing sector and massive investment by large private enterprises. In their recent review of the evidence of dietary transition and food system transformation during economic development, Masters et al. (2022) find that supermarkets have had a rather negative impact on nutrition outcomes, mainly through their association with increased consumption of ultra-processed foods with harmful attributes.

Concrete evidence, however, on the direct effect of supermarkets on consumer nutritional status using detailed micro data is still mixed and context specific. In the US, for example, Drewnowski et al. (2012) find that the risk of obesity is not associated with physical proximity to a supermarket, but it is inversely associated with supermarkets offering higher priced products, concluding that improving access to health foods is beyond improving access to supermarkets. Lear et al. (2013) find that the food basket price of supermarkets is inversely correlated with the body mass index (BMI) of their shoppers. Rogus et al. (2018), in turn, show that the opening of a supermarket in a low-access neighborhood affected local food purchasing and consumption behavior: while the purchase of both healthy and unhealthy foods increased, the consumption of healthy items (water, fruits, vegetables) also increased and of unhealthy items decreased (soft drinks, salty snacks,

pastries). Li et al. (2021) recently performed a systematic review and meta-analysis of 35 studies (most of them in the US and Canada) analyzing access to grocery stores and childhood obesity and find a rather weak relationship, while Zhou et al. (2021) systematic review across 24 studies (21 in the US) reveals a mixed relationship between access to supermarkets and weight-related outcomes.⁶⁰

In developing countries, the literature is similarly not conclusive. Minten and Reardon (2008), using data from Madagascar, find that supermarkets sell higher quality food than traditional markets, but they charge higher prices, even after controlling for quality. Tessier et al. (2008) find that regular users of supermarkets in Tunisia have a slightly improved diet quality. Asfaw (2008) examine the case of Guatemala and show that supermarket purchases increase the share of highly and partially processed foods at the expense of staple foods and are also positively associated with individuals' BMI. Rischke et al. (2015) find that supermarkets in Kenya promote the consumption of processed food, relative to unprocessed food, and increase daily calorie availability. Kimenju et al. (2015) indicate that supermarkets may increase the BMI and probability of overweight among adult consumers in Kenya, but no effects are observed for the BMI among children and adolescents combined with a positive impact on their height-for-age. Umberger et al. (2015) do not find a significant relationship between shopping in supermarkets and nutrition outcomes among adults in Indonesia. Demmler et al. (2018) show that shopping in supermarkets in Kenya contributes to higher consumption of processed and highly processed foods and lower consumption of unprocessed foods, thereby significantly increasing the BMI of the population. In contrast, Debela et al. (2020) find positive effects of supermarket purchases on child nutrition in Kenya.

Overall, the relationship between supermarkets' presence and consumer health and nutrition in urban areas is still not well understood. Part of this ambiguity could be explained by the different driving forces that affect consumer food choices (Ruel et al., 2020), including market competition that has not been encompassed in previous analyses of the linkages between supermarkets and dietary patterns, probably due to the lack of detailed data. While supermarkets may charge, on average, higher or lower food prices or offer a higher or lower share of high-quality foods than

⁶⁰ Adam and Jensen (2016) find, however, that obesity related interventions at retail grocery stores and supermarkets are effective in increasing the purchase of healthy foods.

traditional markets, little has been said about how the level and type of competition they face in specific markets may accentuate (or weaken) these differences, ultimately affecting the dietary habits and health and nutrition of the population, especially the urban poor. The pricing of healthy, nutrient-rich fresh foods versus less healthy processed foods can certainly depend on the level of competition in each area of operation and constitutes an important avenue of future research.⁶¹

4.2 Implications for environmental sustainability

There are important environmental sustainability concerns in the agri-food industry.⁶² According to Tubiello et al. (2021), food systems are responsible for one third of the total global anthropogenic (human-caused) greenhouse gas emissions, where three quarters of these emissions are generated either within the farm gate on in pre- and post-production activities, such as manufacturing, transport, processing, and waste disposal. Crippa et al. (2021) indicate that the largest contribution to greenhouse gas emissions comes from agriculture and land use/land-use change activities. The overuse of chemicals and fertilizers and pesticides has brought environmental problems such as water pollution, biodiversity loss, and nitrous oxide emissions (Eicher and Staatz, 1998). In food retail, the primary energy resources are gas and electricity due to business operations, shopping activities, and equipment used for food preparation, storage, and presentation, which leads to a significant emission of carbon dioxide (Lukic, 2016). In supermarkets, for example, refrigeration, lighting, and heating/air conditioning systems are managed with a high electricity consumption, which has a negative environmental impact (Baldwin, 2011; Beshr et al. 2015). Among the reasons why retailers are hesitant to adopt energysaving technologies is the enormous cost of implementation along with a lack of internal financial resources available to support these changes (Dixon-O'Mara and Ryan, 2017). Food retail operations also contribute to air and water pollution. Most supermarkets discharge grease and food waste directly into municipal sanitary sewers, and vertically integrated food retailers manage warehousing and distribution functions, concentrating air pollutants generated during the production process as well as emissions from trucks in these distribution centers (Davies and

⁶¹ A standard textbook model would predict higher prices with market power, but the impact on nutrition is not clear as higher prices for highly processed or junk food would be good and the opposite for higher fruits and vegetable prices.

⁶² Environmental sustainability refers to the preservation of natural resources by not exceeding the local environment capacity to absorb emissions (Goodland, 1996).

Konisky, 2000; Renko et al., 2010). The increase in concentration in the agri-food system suggests that ecological degradation may be influenced by a small group of major market actors in food and agriculture (Shelke et al., 2011; Hendrickson et al., 2020). Yet, there is little evidence that specifically analyzes the relationship between market concentration and environmental sustainability.

Some of the literature has focused on the upstream segment of the agri-food chain, particularly in the seeds market, and the evidence is mixed. According to the OECD (2018a), researchers have noted that a high degree of market concentration in the seeds market may not necessarily be harmful. Competition among a few large players can in fact promote innovation that contributes to environmental sustainability, productivity, and resilience. However, non-horizontal mergers (i.e., vertical integrations) are not necessarily harmless as they could be used to exclude competitors from accessing important markets or resources. Solberg and Breian (2015), for example, find that consolidation of seed enterprises and vertical integration caused the Nordic countries to concentrate on a few crops and abandon others for nearly fifty years. The shift away from regional and locally produced products decreases diversity, which is detrimental to environmental sustainability. Mustonen-Park (2018) performs a wide-ranging literature review and points that outside of the United States, seed markets are very restricted due to lack of research and development, contributing to a decline of seed varieties.

Several studies have also examined the impact of the food industry on environmental sustainability using the Life Cycle Assessment (LCA) method. This methodology involves incorporating all the material and energy flows associated with a product during its entire life cycle to estimate the environmental impact associated with a food value chain. This kind of research can be useful for identifying which parts of the chain may have an impact on the environment. Yet, since most of the studies are focused on a particular food supply chain and rely on a series of assumptions according to a forthcoming work on the topic by the OECD, they typically lack external validity. Cerutti et. al (2013), Garofalo et al. (2016), Poore and Nemecek (2018), and Shen et al. (2021) are some examples. Most of this literature do not explicitly account for market concentration or competitive practices in these industries but generally find that land use and agricultural production account for most of the total environmental footprint. Keyes et al. (2013)

is one of the few studies that addresses vertical integration through the analysis of environmental performance and opportunities for improvement in apple supply chains in Canada using the life cycle assessment approach. The authors compare the environmental impacts between standard orchard production and cradle-to-retail chains (namely vertically integrated chains) and find that the latter have most adverse effects on electricity for long-term storage and long-distance truck delivery.

Regarding the literature on FLW, there is some discussion of the connection between market concentration and environmental sustainability. FAO (2019) defines FLW as the decrease in quantity or quality of food along the food supply chain.⁶³ In the food supply chain, losses occur during the production, postharvest, storage, transport, and processing stages, whereas food waste is the food discarded in the food retail sector, food service sector, and households (Parfitt et al., 2010; UNEP, 2021). Kummu et al. (2012) estimate that around one quarter of the water resources, cropland, and fertilizers used to produce food are lost due to FLW across the global food supply chain. A study on food waste by European Communities (2011) remarks that environmental sustainability is affected due to transportation and shipping towards landfills combined with upstream production, resources usage, and life cycle processes that could have been avoided. Feedback (2017) argues that food waste is a symptom of structural problems that relate to a power imbalance and lack of incentives to minimize food waste. Power imbalances can result in a decline in innovation and investment that will affect small and medium-sized food businesses in the long term, deriving in higher food prices and fewer choices for consumers. However, there are some authors who argue that waste is part of any optimal production process, and that food waste can be used for other non-food purposes, such as animal feed, fertilizer, and biomass to produce other goods. Under this premise, food waste would be defined only as produce that ends up in landfills (Bellemare et al., 2017).

In this context, there are only few studies examine FWL in specific food value chains and their links with market concentration and/or vertical integration. Some of these focus on developed countries and find that most of the losses stem from the power imbalance in the retail side of the

⁶³ Quantitative losses refer to food that exits the food supply chain, while qualitative losses refer to the decrease in food attributes that reduces its value for intended use (Cattaneo et al., 2021)

chain. Devin and Richardson (2018), for example, assesses how retailers in the fresh fruit supply chain in Australia contribute significantly to food waste by interviewing different actors along the chain, including retailers and producers. Since the bargaining power of retailers is high, they can pass food waste responsibility upward the chain to growers by offering lower prices after delivery and setting strict standards regarding the cosmetic appearance of food that leads to discard suitable food. Ghosh and Erickson (2019) use five-year company data for a medium-sized premium bread manufacturer and supplier in Sweden to examine how take-back agreements can be used by retailers to exert coercive control over suppliers in highly concentrated markets. In these contracts, the retailer is only required to pay the supplier for the quantity purchased but not for the quantity ordered. As a result, larger downstream players can negotiate better contracts with producers and reject viable products that end up being thrown away. Herzberg (2022) examines fresh fruit and vegetable supply chains in Germany based on multiple interviews with stakeholders. The author notes that the industry exhibits an upward trend of concentration and consolidation that are associated with short-notice cancellations, order changes, and sudden changes in quality requirements that contribute to food losses and waste.

Studies in developing countries, where market concentration and vertical integration are not that high, present mixed evidence regarding how market concentration and/or participation in modern commercialization channels influence FWL. Minten et al. (2020), for instance, explore postharvest losses (PHL) experienced by teff milk and perishable liquid milk in rural and urban areas of Ethiopia. The authors analyze PHL at all levels of the chain and conclude that modern retailers generate lower PHL than traditional retailers. More specifically, modern retailers maintain higher quality standards in the procurement process, provide better packaging and protection systems, and provide refrigeration, storage, and sales facilities that offer better quality. However, the authors argue that the stricter requirements by modern retail could also contribute to larger PHL upstream. Chanboud and Moustier (2020), in turn, examine FLW in the tomato supply chain (from production to retail), comparing the amount generated by supermarkets and non-supermarket channels in Colombia. The authors do not find significant differences in the resulting FLW between both commercialization channels.

All in all, the links between market concentration and environmental sustainability can be studied from different perspectives as vertical and horizontal integrations that favor concentration may occur both upstream and downstream the food value chain, which can result in varying competitive practices and environmental consequences. Yet, the current evidence is highly context specific as it pertains to specific market segments, supply chains, or regions.⁶⁴ In this sense, there is an important research gap that could be fulfilled by studying potential environmental effects across different value chains and market segments that are important for the food industry, as well as on the efficient use of resources such as water, land, and energy. The scarce literature in developing regions is also evident, where the lack of concentration and/or vertical integration could alternatively be a cause of food waste as smallholder farmers typically do not have the incentives or means to adopt food loss preventive practices, while downstream coordination problems with small-scale operating agents, who additionally lack of sufficient storage, transportation, and processing capacities, could favor food losses.

⁶⁴ Related to this matter, Barrett et al. (2021) remark that firms continuously search for new markets, more efficient management practices, inputs of higher quality, and new ways to differentiate their products in order to generate new profit sources as their past innovations diffuse to competitors. A study on sustainability and competition by the OECD (2020) points out, in turn, that the answer to the well-known Schumpeter-Arrow debate regarding whether monopoly or competition leads to more innovation, and in particular innovation in sustainability, can be better assessed with concrete situations.

5. Concluding remarks and the way forward

This study attempted to provide a comprehensive assessment of recent trends in market concentration in the agri-food industry and on how and to what extent concentration could affect market conduct and performance of food systems, as well as implications for food security and nutrition and environmental sustainability.

Regarding the current state of market concentration in the agri-food industry, the upstream segment of the chain is composed of highly concentrated industries, including fertilizers, agrochemicals, and seeds, which appear to be in a consolidation path with important global market players. Markets midstream the chain, particularly food manufacturing and processing, as well as downstream, especially supermarket retailing, similarly exhibit moderate to high levels of concentration that have either increased or remained stable in recent years across several countries. Grain commodity trade is also highly concentrated. Still, important data gaps remain across industries and regions that prevent a more comprehensive picture of market structure (concentration) in agri-food systems, including difficulties in accessing proprietary market and scientific data and market share data at more disaggregated levels, as well as analyses that are typically geographically narrowed. In addition, there are data aggregation challenges and complications, including the use of different variables for analysis (e.g., shares based on volume versus sales) or defining the relevant markets across segments. These shortcomings point to the necessity of devoting more efforts on data access, compilation, and processing.

Market concentration and consolidation along the chain over the past years have been largely led by mergers and acquisitions combined with the transformation of agri-food chains, including the expansion of supermarket retailing worldwide. While these increasing horizontal and vertical integrations can be associated with higher efficiencies (particularly through economies of scale and better coordination), they could also result in market power exertion, tacit collusion, and other anticompetitive practices in the absence of these efficiencies and other offsetting factors. Recent events, including the global pandemic and the Russian aggression against Ukraine, with disruptions in agri-food chains and record high (input) prices, have further raised these concerns. Yet, the relationship between concentration and market power exertion is multifaceted and the evidence supporting market power abuse or anticompetitive behavior is not generally obvious and may be context specific. Concentration alone is not sufficient to exert market power or to effectively coordinate strategies among few large market players as this also depends on other market characteristics and factors, including the type and nature of competition. Deriving conclusive evidence regarding market power exertion is further challenging due to lack of comprehensive and detailed data.

Most reviews of market power in the agri-food industry have mainly focused on midstream and downstream markets in developed economies, where there is more data availability, and do not find strong evidence of market power exertion, particularly buyer power from food processing firms. The evidence on asymmetric price transmission and its relationship with buyer concentration is similarly not conclusive. In developing regions, the literature shows the importance of contextual factors such as remoteness, financial liquidity constraints, credibility, and reputation issues in conferring considerable market power; high transaction costs (search, storage, contracting costs) can also sharply limit competition as well as intra-national trade frictions. In these contexts, intermediaries can take large surpluses in detriment of consumers and farmers. Upstream, high levels of concentration in input markets, combined with elevated levels of multimarket contact in those industries, are a matter of further investigation in terms of the effects that reduced competition or coordinated actions could have on agricultural productivity. Additionally, more research is required to assess the impact of unfair trading practices (UTPs) along the chain (not only on farmers), while efficiency and rent distributional implications of some practices are still not fully clear.

New approaches and methodologies combined with access to more disaggregated data can be helpful for an improved understanding of the complex relationship between market concentration, conduct, and performance at various points in the agri-food chain and better inform eventual policy decisions. Some of these new approaches include an emerging strand of research that frames vertical relationships between suppliers and retailers in terms of bargaining power instead of the traditional market power approach; recent developments in markup estimation methods; and novel experimental approaches on market concentration and power and their welfare implications, particularly in less developed countries, although the results may not be generalizable. In developing countries, as more data (including scanner data) become available, the implications of the transformation of food value chains (including favoring concentration) on market performance could also be studied relying on advanced empirical industrial organization methods that have been widely used in developed economies.

On the implications of market concentration on food security and nutrition, the association between participation in modern commercialization channels, especially supermarket channels, and access to better diets among small farmers seems to be positive, but the literature is still relatively limited to derive more generalized conclusions. The impacts could be affected by a variety of factors, such as contract terms and farmers' conditions, and more research is required to explore these factors and extend the research to other regions (beyond Africa and South Asia). Additionally, future studies should attempt to capture more detailed data, such as macro and micronutrient data, to evaluate effects beyond consumption of specific food groups and dietary diversity. Future research about this subject could explore some of the following specific questions: Does participation in modern commercialization channels, especially supermarket channels, promote access to better and healthier diets to all participating farm households or are these impacts heterogenous or specific to particular groups? Do changes in dietary patterns hold in the medium and long term? What are the impacts on health and nutritional outcomes among household members, especially women and children? Do the effects observed in Africa or South Asia replicate in other developing regions? Do market conditions affect the contractual arrangements of smallholders participating in modern marketing channels and thereby their food access and diet quality? These questions can help to better understand the implications of participating in modern commercialization channels on farm household health and nutrition and aid in the design of policies that can enhance the potential benefits of smallholders' participation in modern integrated markets.

In a similar vein, the relationship between supermarkets' presence and consumer health and nutrition is still not well understood. This could be explained by the different factors that influence consumer food choices, including market competition that has not been accounted for in previous analyses of the linkages between supermarkets and dietary patterns, probably due to data restrictions. Food prices and quality may, on average, be higher or lower in supermarkets compared

to traditional markets, but these differences may be aggravated or lessen depending on the degree of competition faced by supermarkets. More important, the pricing of healthy, nutrient-rich fresh foods versus less healthy processed foods may also depend on the level of competition and can have important implications for the health and nutrition of the population, especially the urban poor. Specific questions that could be addressed in future work include: Do the price differentials between the set of healthy and less healthy foods vary with the level of competition faced by supermarkets? Are supermarkets more likely to change their pricing behavior when they directly compete with other supermarket chains than when they compete with traditional markets? How do changing market conditions and consequent varying relative pricing behavior of supermarkets affect consumer food choices and the health and nutrition of the population, especially the poor? How do changing market conditions affect non-price competition in terms of product quality and information, including labeling and stop signs, and how can public policies contribute to influence those decisions? This type of analysis will allow to better inform how institutional arrangements and regulatory framework governing market activity in the retail sector could be modified to improve market performance (if applicable) and promote the consumption of healthier foods, particularly among vulnerable populations.

On the implications of market concentration on environmental sustainability, the literature on the links between concentration and environmental sustainability is highly context specific as it pertains to particular market segments, supply chains, or regions. In the seeds sector, for example, the evidence is mixed. Similarly, the literature on food loss and waste (FLW) points out that food waste could be a symptom of structural problems that relate to power imbalances and lack of incentives to minimize food waste. Available studies focusing on developed countries and specific food value chains argue that most of the losses stem from the power imbalance in the retail side of the chain. Certainly, in developing countries the lack of concentration and/or vertical integration could also be a cause of food waste as smallholders generally do not have the incentives or resources to implement food loss preventive practices, while downstream coordination difficulties with small-scale operating agents, who further have limited storage, transportation, and processing capacities, could contribute to food losses; this hypothesis, however, has not been much explored in the literature. Some research questions that may be interesting to further explore in the future include: *How is concentration upstream and downstream the agri-food chain affecting the*

environmental sustainability of two key agricultural resources such as water and land? Are more concentrated markets being less or more efficient in energy use with more advanced processing, cooling, packaging, and other energy-efficient practices? Is market concentration along the agrifood chain favoring specific competitive behavior and practices that, for example, hinder technology development and research or promote FWL? Is the extent of the situation on this regard different in developed versus developing countries?

In sum, the relationship between market concentration, conduct, and performance should be regarded as complex where the occurrence of mark-up pricing, unfair trading practices, collusion, crowding out of small-scale businesses, lower access to healthy diets, less innovation, and more (or less) food waste may be context-specific and influenced by factors other than concentration. As pointed above, several key data and research gaps remain to better understand these relationships and derive more generalized conclusions.

An action that could be implemented to alleviate part of the data and research gaps could be to expand competition authorities' capabilities to perform more in-depth market studies in specific industries or sectors along the agri-food value chain and encourage the disclosure of more detailed (and frequent) data. More important, it is key to invest in stronger national and multinational competition authorities and encourage more coordination among them to facilitate the convergence and effective enforcement of antitrust laws at the local, regional, and global level, and ensure that regulations and other policies do not accidentally create barriers to entry, restrain competition, or restrict trade between countries. The International Competition Network (ICN) is an interesting initiative on this regard as it is a global body devoted exclusively to competition law enforcement and its members represent national and multinational competition authorities. Yet, it is equally important to keep in mind that guaranteeing competition among agri-food markets will not solve poverty, food insecurity and malnutrition, and environmental issues, such that other, more targeted policies are almost certainly needed.

Similarly, with the wide use of new information technologies, the development of easily accessible virtual platforms (whether through local, regional, or global initiatives) could also contribute to increase market transparency and efficiency across different agri-food industries. These platforms

could report relevant market data, promote business (trade) opportunities and access to new markets, or even permit to submit complaints about potential regulatory violations or unfair trading practices.

References

Adam, A. & Jensen, J.D. (2016). What is the effectiveness of obesity related interventions at retail grocery stores and supermarkets? - A systematic review. *BMC Public Health*, Dec 28; 16(1):1247.

Andersson, C.I.M., Chege, C.G.K., Rao, E.J.O. & Qaim, M. (2015). Following up on smallholder farmers and supermarkets in Kenya. *American Journal of Agricultural Economics* 97(4): 1247–1266.

AMS-USDA. (2015). Report for Cotton Varieties Planted. Agricultural Marketing Service, Cotton and Tobacco Program, USDA.

AMS-USDA. (2022). Competition in Food Retail and Distribution Markets and Access for Agricultural Producers and Small and Midsized Food Processors. Agricultural Marketing Service, USDA. *Federal Register*, Vol. 87, No. 52.

Asfaw, A. (2008). Does supermarket purchase affect the dietary practices of households? Some empirical evidence from Guatemala. *Development Policy Review* 26(2): 227–243.

Atkin, D. & Donaldson, D. (2015). Who's Getting Globalized? The Size and Implications of Intranational Trade Costs. NBER Working Paper No. 21439.

Baffes, J. & Koh, W.C. (2022). Fertilizer prices expected to remain higher for longer. *World Bank Blogs*, May 11, 2022. Available at: <u>https://blogs.worldbank.org/opendata/fertilizer-prices-expected-remain-higher-longer</u>

Baker, J. B. & Farrell, J. (2019). Oligopoly coordination, economic analysis, and the prophylactic role of horizontal merger enforcement. *University of Pennsylvania Law Review* 168: 1985.

Baldwin, C. (2011). Sustainability in Food Retailing. In C. Baldwin, *Sustainability in the Food Industry* (pp. 213-224). Institute of Food Technologists Series. Barrett, C. B. (1997). Food marketing liberalization and trader entry: evidence from Madagascar. *World Development* 25 (5): 763–777.

Barrett, C.B., Bachke, M.E., Bellemare, M.F., Michelson, H.C., Narayanan, S. & Walker, T.F. (2012). Smallholder participation in contract farming: Comparative evidence from five countries. *World Development* 40(4): 715–730.

Barrett, C., Reardon, T., Swinnen, J. & Zilberman, D. (2019). Structural transformation and economic development: Insights from the agri-food value chain revolution. Mimeo, Dyson School of Applied Economics and Management, Cornell University.

Barrett, C., Reardon, T., Swinnen, J. & Zilberman, D. (2021). Agri-food value chain revolutions in low- and middle-income countries. *Journal of Economic Literature*, forthcoming.

Begemann, S. (2015). Seed competition heats up. Farm Journal Magazine. Available at: <u>https://www.verdantpartners.com/post/seed-competition-heats-up</u>.

Bellemare, M.F. (2012). As you sow, so shall you reap: The welfare impacts of contract farming. *World Development* 40(7): 1418–1434.

Bellemare, M.F., Cakir, M., Peterson, H.H., Novak, L. & Rudi, J. (2017). On the Measurement of Food Waste. *American Journal of Agricultural Economics* 99(5): pp. 1148–1158.

Berry, S., Gaynor, M. & Scott Morton, F. (2019). Do Increasing Markups Matter? Lessons from Empirical Industrial Organization. *Journal of Economic Perspectives* 33(3): pp. 44–68.

Bernheim, B. and M. Whinston (1990), Multimarket Contact and Collusive Behavior. *The RAND Journal of Economics* 21(1): 1-26.

Bergquist, L. F. & Dinerstein, M. (2020). Competition and Entry in Agricultural Markets: Experimental Evidence from Kenya. *American Economic Review* 110(12): 3705–47.

Beshr, M., Aute, V., Sharma, V. & Radermacher, R. (2015). A comparative study on the environmental impact of supermarket refrigeration systems using low GWP. *International Journal of Refrigeration* 56: 154–164.

Bjornlund, V., Bjornlund, H. & Van Rooyen, A.F. (2020). Why agricultural production in sub-Saharan Africa remains low compared to the rest of the world – a historical perspective. *International Journal of Water Resources Development* 36(sup 1): S20-S53.

Bonanno, A., Russo, C. & Menapace, L. (2018). Market power and bargaining in agrifood markets: A review of emerging topics and tools. *Agribusiness* 34(1): 6–23.

Bonny, S. (2014). Taking stock of the genetically modified seed sector worldwide: market, stakeholders, and prices. *Food Security* 6(4): 525–540.

Bonny, S. (2017). Corporate concentration and technological change in the global seed industry. *Sustainability* 9(9): 1632.

Bonnet, C., & Bouamra-Mechemache, Z. (2016). Organic Label, Bargaining Power, and Profit-Sharing in the French Fluid Milk Market. *American Journal of Agricultural Economics* 98(1): 113–33.

Bonroy, O., & Constantatos, C. (2015). On the economics of labels: How their introduction affects the functioning of markets and the welfare of all participants. *American Journal of Agricultural Economics* 97(1): 239–259.

Bresnahan, T. (1989). Chapter 17. Empirical Studies of Industries with Market Power. *The Handbook of Industrial Organization*, vol 2. Richard Schmalensee and Robert Willig, eds., North Holland, pp. 1011-1057.

Bryant, H., Maisashvili, A., Outlaw, J. & Richardson, J. (2016). Effects of Proposed Mergers and Acquisitions Among Biotechnology Firms on Seed Prices, AFPC Working Paper, No. 16-2, Texas A&M University. Available at:

https://www.afpc.tamu.edu/research/publications/files/675/WP 16-2.pdf

Busso, M. & Galiani, S. (2019). The causal effect of competition on prices and quality: Evidence from a field experiment. *American Economic Journal: Applied Economics* 11(1): 33-56.

Çakır, M. & Nolan, J. (2015). Revisiting Concentration in Food and Agricultural Supply Chains: The Welfare Implications of Market Power in a Complementary Input Sector. *Journal of Agricultural and Resource Economics* 40(2): 203–219.

Çakır, M., Kong, X., Cho, C. & Stevens, A. (2020). Rural Food Retailing and Independent Grocery Retailer Exits. *American Journal of Agricultural Economics* 102(5): 1352-1367.

Carletto, C., Corral, P. & Guelfi, A. (2017). Agricultural commercialization and nutrition revisited: Empirical evidence from three African countries. *Food Policy* 67: 106–118.

Casaburi, L. & Macchiavello, R. (2019). Demand and supply of infrequent payments as a commitment device: evidence from Kenya. *American Economic Review* 109(2): 523–555.

Casaburi, L., & Reed, T. F. (2017). Competition in Agricultural Markets: An Experimental Approach. *CEPR Discussion Paper* no. DP11985.

Casaburi, L., & Reed, T. F. (2022). Using Individual-Level Randomized Treatment to Learn about Market Structure. *American Economic Journal: Applied Economics*, (forthcoming).

Cattaneo, A., Sánchez, M.V., Torero, M. & Vos, R. (2021). Reducing food loss and waste: Five challenges for policy and research. *Food Policy* 98: 101974.

Cerutti, A. K., Bruun, S., Donno, D., Beccaro, G.L. & Bounous, G. (2013). Environmental sustainability of traditional foods: the case of ancient apple cultivars in Northern Italy assessed by multifunctional LCA. *Journal of Cleaner Production* 52: 245–252.

Chaboud, G. & Moustier, P. (2020). The role of diverse distribution channels in reducing food loss and waste: The case of the Cali tomato supply chain in Colombia. *Food Policy* 98: 101881.

Chege, C.G.K., Andersson, C.I.M. & Qaim, M. (2015). Impacts of supermarkets on farm household nutrition in Kenya. *World Development* 72: 394–407.

Cheng, I.H. & Xiong, W. (2014). Financialization of commodity markets. *Annual Review of Financial Economics* 6: 419–441.

Crespi, J. & MacDonald, J. (2022). Concentration in food and agricultural markets. *Handbook of Agricultural Economics*, Vol. 6, Ch. 87, 4781-4843.

Crippa, M., Solazzo, E., Guizzardi, D., Monforti-Ferrario, F., Tubiello, F.N. & Leip, A. (2021). Food systems are responsible for a third of global anthropogenic GHG emissions. *Nature Food* 2: 198-209.

Davies, T. & Konisky, D.M. (2000). *Environmental implications of the food services and food retail industries*. Resources for the Future, Discussion Paper.

De Loecker, J., Eeckhout, J. & Unger G. (2020). The Rise of Market Power and the Macroeconomic Implications. *The Quarterly Journal of Economics* 135(2): 561–644, <u>https://doi.org/10.1093/qje/qjz041</u>

De Loecker J., Warzynski, F. M. P. (2012). Markups and Firm-Level Export Status, *American Economic Review* 102 (6): 2437–2471.

Debela, B.L, Demmler, K.M., Klasen, S. & Qaim, M. (2020). Supermarket food purchases and child nutrition in Kenya. *Global Food Security* 25: 1000341.

Debela, B.L., Ruml, A. & Qaim, M. (2022). Effects of contract farming on diets and nutrition in Ghana. *Applied Economic Perspectives and Policy* 44(2): 911–929.

Deconinck, K. (2019). New evidence on concentration in seed markets. *Global Food Security* 23: 135–138.

Deconinck, K. (2020). Concentration in seed and biotech Markets: Extent, causes, and impacts. *Annual Review of Resource Economics* 12(1): 129–147.

Deconinck, K. (2021). Concentration and market power in the food chain. OECD Food, Agriculture and Fisheries Papers No. 151, OECD Publishing, Paris. http://dx.doi.org/10.1787/3151e4ca-en

Demmler, K.M., Ecker, O. & Qaim, M. (2018). Supermarket shopping and nutritional outcomes: A panel data analysis for urban Kenya. *World Development* 102: 292–303.

Demsetz, H. (1973). Industry Structure, Market Rivalry and Public Policy. *Journal of Law and Economics* 16(1): 1–9.

Devin, B. & Richards, C. (2018). Food Waste, Power, and Corporate Social Responsibility in the Australian Food Supply Chain. *Journal of Business Ethics* 150(1): 199–210.

Di Marcantonio, F., P. Ciaian & Castellanos V. (2018). Unfair trading practices in the dairy farm sector: Evidence from selected EU regions. *JRC Technical Reports*, European Commission, http://dx.doi.org/10.2760/747043.
Di Marcantonio, F., P. Ciaian & Falkowski, J. (2020). Contracting and Farmers' Perception of Unfair Trading Practices in the EU Dairy Sector. Journal of Agricultural Economics 71(3): 877-903.

Dillon, B. & Dambro, C. (2017). How competitive are crop markets in Sub-Saharan Africa? American Journal of Agricultural Economics 99(5): 1344–1361.

Dixon-O'Mara, C. & Ryan, L. (2018). Energy efficiency in the food retail sector: barriers, drivers and acceptable policies. Energy Efficiency 11: 445-464.

Dobson, P.W., Waterson, M. & Davies, S.W. (2003). The Patterns and Implications of Increasing Concentration in European Food Retailing. Journal of Agricultural Economics 54(1): 111-125.

Drewnowski, A., Aggarwal, A., Hurvitz, P.M., Monsivais, P. & Moudon, A.V. (2012). Obesity and supermarket access: Proximity or price? American Journal of Public Health 102(8): 74-80.

Ebata, A. & Hernández, M.A. (2017). Linking smallholder farmers to markets on extensive and intensive margins: Evidence from Nicaragua. Food Policy 73: 34-44.

Eicher, C.K. & Staatz, J.M. (1998). International Agricultural Development. Johns Hopkins University Press.

Einarsson, A. (2007). The retail sector in the Nordic countries: A comparative analysis. Bifröst Journal of Social Science 1: 31–42.

ERS-USDA. (2021a). Retail Trends, Economic Research Service, USDA. Available at https://www.ers.usda.gov/topics/food-markets-prices/retailing-wholesaling/retail-trends/.

ERS-USDA. (2021b). Wholesaling, Economic Research Service, USDA. Available at https://www.ers.usda.gov/topics/food-markets-prices/retailing-wholesaling/wholesaling/.

European Commission. (2014). The Economic Impact of Modern Retail on Choice and Innovation in the EU Food Sector. Available at:

https://ec.europa.eu/competition/sectors/agriculture/retail study report en.pdf

European Commission. (2019). Antitrust: Commission opens investigation into possible collusion by two French retailers in a purchasing alliance. Brussels.

European Communities. (2011). Preparatory Study on Food Waste Across EU 27.

Falkowski, J. (2017). The Economic Aspects of Unfair Trading Practices: Measurement and Indicators, in Unfair Trading Practices in the Food Supply Chain: A Literature Review on Methodologies, Impacts and Regulatory Aspects, Di Marcantonio, F. & Ciaian P. (eds.).

FAO. (2019). State of Food and Agriculture 2019. Moving forward on food loss and waste reduction.

FAS-USDA. (2022). Grain: World Markets and Trade. Foreign Agricultural Service (FAS), United States Department of Agriculture, June. Available at: <u>https://www.fas.usda.gov/data/grain-world-markets-and-trade</u>.

Feedback & Rockefeller Foundation. (2017). *Causes of Food Waste in International Value Chains*. Feedback.

Fernandez, N. & Marin, P.L. (1998). Market Power and multimarket contact: Some evidence from the Spanish hotel industry. *The Journal of Industrial Economics* 46(3): 301-315.

Fernandez-Cornejo, J. (2004). The seed industry in U.S. agriculture: an exploration of data and information on crop seed markets, regulation, industry structure, and research and development. *Agric. Inf. Bull.* 786, USDA, Washington, DC. Available at: https://www.ers.usda.gov/webdocs/publications/42517/13616 aib786 1 .pdf

Fernandez-Cornejo, J. & Just, R. (2007). Researchability of Modern Agricultural Input Markets and Growing Concentration. *American Journal of Agricultural Economics* 89(5): 1269–1275.

Fraas, A.G. & Greer, D.F. (1977). Market structure and price collusion: An empirical analysis. *The Journal of Industrial Economics* 26(1): 21–44.

Fortune Business Insights. (2022). Fast Food Market Size, Share & COVID-19 Impact Analysis, By Product Type (Burger & Sandwich, Pizza & Pasta, Asian & Latin American Food, and Others), Service Type (On-Premise and Delivery & Take Away), and Regional Forecast, 2021-2028. Report ID FBI106482.

Fuglie, K. O., Heisey, P. W., King, J.L., Pray, C.E., Day-Rubenstein, K., Schimmelpfennig, D., Wang, S.L. & Karmarkar-Deshmukh, R. Research Investments and Market Structure in the Food Processing, Agricultural Input, and Biofuel Industries Worldwide. ERR-130. Economic Research Service, USDA.

Garofalo, P., D'Andrea, L., Tomaiuolo, M., Venezia, A. & Castrignanò, A. (2017). Environmental sustainability of agri-food supply chains in Italy: The case of the whole-peeled tomato production under life cycle assessment methodology. *Journal of Food Engineering* 200: 1–12.

Ghosh, R. & Eriksson, M. (2019). Food waste due to retail power in supply chains: Evidence from Sweden. *Global Food Security* 20: 1–8.

GIPSA-USDA. (2016). Statistical Report and Annual Report, 2016. Grain Inspection, Packers and Stockyards Administration, USDA.

Glauber, J. & Laborde, D. (2022) How will Russia's invasion of Ukraine affect global food security? *IFPRI Blog: Issue Post.* February 24.

Gomez, M.I. & Ricketts, K.D. (2013). Food value chain transformations in developing countries: Selected hypotheses on nutritional implications. *Food Policy* 42: 139–150.

Goodland, R. & Daly, H. (1996). Environmental Sustainability: Universal and Non-Negotiable. *Ecological Applications* 6(4): 1002–1017.

Haggblade, S., Minten, B., Pray, C., Reardon, T. & Zilberman, D. (2017). The herbicide revolution in developing countries: Patterns, causes, and implications. *The European Journal of Development Research* 29(3): 533–559.

Hamilton, S. F., Liaukonyte, J. & Richards, T. J. (2020). Pricing Strategies of Food Retailers. *Annual Review of Resource Economics* 12(1): 87–110.

Hausmann, J. & Leibtag, E. (2007). Consumer benefits from increased competition in shopping outlets: Measuring the effect of Wal-Mart. *Journal of Applied Econometrics* 22(7): 1157–77.

Hawkes, C. (2008). Dietary Implications of Supermarket Development: A Global Perspective. *Development Policy Review* 26(6): 657–692.

Hebebrand, C. & Laborde, D. (2022). High fertilizer prices contribute to rising global food security concerns. *IFPRI Blog: Issue Post*, Apr 25, 2022. Available at: <u>https://www.ifpri.org/blog/high-fertilizer-prices-contribute-rising-global-food-security-concerns</u>

Hendrickson, M., Howard, P. H., Miller, E. M. & Constance, D. (2020). *The food system: concentration and its impacts*. Family Farm Action Alliance.

Henson, S. (2006). The Role of Public and Private Standards in Regulating International Food Markets. Paper prepared for the IATRC Summer symposium "Food Regulation and Trade: Institutional Framework, Concepts of Analysis and Empirical Evidence."

Hernández, M.A. & Torero, M. (2018). Promoting Competition in the Fertilizer Industry in Africa. A Global and Local Approach. *IFPRI Issue Brief*, March.

Hernández, M.A. & Torero, M. (2013). Market concentration and pricing behavior in the fertilizer industry: A global approach. *Agricultural Economics* 44(6): 723–734.

Hernández, M.A. & Torero, M. (2011). Fertilizer market situation: Market structure, consumption and trade patterns, and pricing behavior. IFPRI Discussion Paper #1058.

Herrmann, R., Nkonya, E., & Faße, A. (2018). Food value chain linkages and household food security in Tanzania. *Food Security* 10(4): 827–839.

Herzberg, R., Schmidt, T. & Keck, M. (2022). Market power and food loss at the producer-retailer interface of fruit and vegetable supply chains in Germany. *Sustainability Science* 1: 1–15.

Holm, T., Loy, J. P., & Steinhagen, C. (2012). Cost pass-through in differentiated product markets: A disaggregated study for milk and butter. In *Transparency of Food Pricing TRANSFOP* (No. Grant Agreement No. KBBE-265601-4-TRANSFOP). <u>https://doi.org/10.1093/erae/jbu031</u>

Horn, H. & Wolinsky, A. (1988). Bilateral monopolies and incentives for merger. *The RAND Journal of Economics* 19(3): 408–419.

Hovenkamp, H. J. & Shapiro, C. (2018). Horizontal Mergers, Market Structure, and Burdens of Proof. *Yale Law Journal* 127(7): 1996-2025.

Howard, P. (2016). Concentration and Power in the Food System: Who Controls What We Eat. New York: Bloomsbury Publishing.

IPES-Food (2017). Too big to feed: Exploring the impacts of mega-mergers, concentration, concentration of power in the agri-food sector. International Panel of Experts on Food Systems, October.

Ji, I., Chung, C. & Lee, J. (2017). Measuring Oligopsony Power in the U.S. Cattle Procurement Market: Packer Concentration, Cattle Cycle, and Seasonality. *Agribusiness* 33(1): 16–29.

Jia, P. (2008). What happens when Wal-Mart comes to town: An empirical analysis of the discount retailing industry. *Econometrica* 76(6): 1263–1316.

Just, R. E., & Hueth, D. L. (1993). Multimarket exploitation: The case of biotechnology and chemicals. *American Journal of Agricultural Economics* 75(4): 936–945.

Kaditi, E. (2012). Market dynamics in food supply chains: The impact of globalization and consolidation on firms' market power. *Agribusiness* 29(4): 410–425.

Keyes, S. (2013). Evaluating the Environmental Impacts of Conventional and Organic Apple Production in Nova Scotia, Canada, Through Life Cycle Assessment. Dalhousie University.

Kim, C., Hallahan, C., Taylor, H. & Schluter, G. (2002). Market power and cost-efficiency effects of the market concentration in the U.S. nitrogen fertilizer industry. Paper prepared for the 2002 Annual Meeting of the American Agricultural Economics Association.

Kimenju, S.C., Rischke, R., Klasen, S. & Qaim, M. (2015). Do supermarkets contribute to the obesity pandemic in developing countries? *Public Health Nutrition* 18(17): 3224–3233.

Kolasky, W.J. & Dick, A.R. (2003). The Merger Guidelines and the Integration of Efficiencies into Antitrust Review of Horizontal Mergers. *Antitrust Law Journal* 71(1): 207-251. Available at: <u>https://www.justice.gov/archives/atr/merger-guidelines-and-integration-efficiencies-antitrust-review-horizontal-mergers</u>

Koppenberg, M. & Hirsch, S. (2021). Markup estimation: A comparison of contemporary methods at the example of European food retailers. *Agribusiness* 38(1): 108–133.

Kummu, M., de Moel, H., Porkka, M., Siebert, S., Varis, O. & Ward, P.J. (2012). Lost food wasted resources: global food supply chain losses and their impacts on freshwater, cropland, and fertiliser use. *Science of The Total Environment* 438: 477–489.

Kwoka, J. (2016). The structural presumption and the safe harbor in merger review: False positives or unwarranted concerns. *Antitrust Law Journal* 81(3): 837–872.

Lear, S.A., Gasevic, D. & Schuurman, N. (2013). Association of supermarket characteristics with the body mass index of their shoppers. *Nutrition Journal* 12:117. doi: 10.1186/1475-2891-12-117.

Lee, A., Mhurchu, C.N., Sacks, G., Swinburn, B., Snowdon, W., Vandevijvere, S., Hawkes, C., L'abbé, M., Rayner, M., Sanders, D., Barquera, S., Friel, S., Kelly, B., Kumanyika, S., Lobstein, T., Ma, J., Macmullan, J., Mohan, S., Monteiro, C., Neal, B. Walker, C. & INFORMAS (2013). Monitoring the price and affordability of foods and diets globally. *Obesity Reviews* 14 Suppl 1:82–95.

Lee, H. and Van Cayseele, P. (2022). Market power and the volatility of markups in the food value chain: The role of Italian cooperatives. *European Review of Agricultural Economics*, forthcoming.

Li, Y., Luo, M., Wu, X., Xiao, Q., Luo, J. & Jia, P. (2021). Grocery store access and childhood obesity: A systematic review and meta-analysis. *Obesity Reviews* 22 Suppl 1: e12945.

Lloyd, T. (2017). Forty years of price transmission research in the Food Industry: Insights, challenges and prospects. *Journal of Agricultural Economics* 68(1): 3–21.

Lukic, R. (2016). Energy efficiency in the food retail. *Business Excellence and Management* 6(1): 16–35

Macchiavello, R. & Morjaria A. (2021). Competition and relational contracts in the Rwanda coffee chain. *Quarterly Journal of Economics* 136(2): 1089–1143.

Macchiavello, R., Reardon, T. & Richards, T.J. (2022). Empirical industrial organization economics to analyze developing country food value chains. *Annual Review of Resource Economics*, 14, forthcoming.

MacDonald, J. (2017), Consolidation, Concentration, and Competition in the Food System, *Economic Review (Federal Reserve Bank of Kansas City)* 102: 85–105.

Mammana, I. (2014), Concentration of market power in the EU seed market. Available at: <u>https://www.greens-</u>efa.eu/files/assets/docs/concentration of market power in the eu seed market.pdf

Masters, W.A., Finaret, A.B. & Block, S.A. (2022). The economics of malnutrition: Dietary transition and food system transformation. Handbook of Agricultural Economics. Elsevier, <u>https://doi.org/10.1016/bs.hesagr.2022.03.006</u>

McCorriston, S. (2014). Background Note, in OECD (ed.), *Competition Issues in the Food Chain Industry*, OECD Publishing, Paris.

McCorriston, S. (ed.) (2015). Price Transmission in Food Markets, Oxford University Press. Moser, C., Barrett, C. B. & Minten, B. (2009). Spatial integration at multiple scales: rice markets in Madagascar. *Agricultural Economics* 40(3): 281–294.

Mérel, P. & Sexton, R.J. (2017). Buyer power with atomistic upstream entry: Can downstream consolidation increase production and welfare? *International Journal of Industrial Organization* 50: 259–293.

Michelson, H.C. (2013). Small farmers, NGOs, and a Walmart world: Welfare effects of supermarkets operating in Nicaragua. *American Journal of Agricultural Economics* 95(3): 628–649.

Miller, N., Berry, S., Scott Morton, F., Baker, J., Bresnahan, T., Gaynor, M., Gilbert, R., Hay, G., Jin, G., Kobayashi, B., Lafontaine, F., Levinsohn, J., Marx, L., Mayo, J., Nevo, A., Pakes, A., Rose, N., Rubinfeld, D., Salop, S., Schwartz, M, Seim, K., Shapiro, C., Shelanski, H., Sibley, D., Sweeting, A. & Wosinska, M. (2022). On the misuse of regressions of price on the HHI in merger review. *Journal of Antitrust Enforcement* 10: 248–259

Minten, B. & Reardon, T. (2008). Food Prices, Quality, and Quality's Pricing in Supermarkets versus Traditional Markets in Developing Countries. *Review of Agricultural Economics* 30(3): 480–490.

Minten, B., Tamru, S. & Reardon, T. (2020). Post-harvest losses in rural-urban value chains: Evidence from Ethiopia. *Food Policy* 98: 101860.

Mishra, A.K., Kumar, A., Joshi, P.K. & D'Souza, A. (2018). Production Risks, Risk Preference and Contract Farming: Impact on Food Security in India. *Applied Economic Perspectives and Policy* 40(3): 353–378.

Miyata, S., Minot, N. & Hu, D. (2009). Impact of contract farming on income: Linking small farmers, packers, and supermarkets in China. *World Development* 37(11): 1781–1790.

Monteiro, C.A., Moubarac, J-C., Cannon, G., Ng, S.W. & Popkin, B. (2013). Ultra-processed products are becoming dominant in the global food system. *Obesity Reviews* 14(2): 21–28.

Moser, C., Barrett, C. B. & Minten, B. (2009). Spatial integration at multiple scales: rice markets in Madagascar. *Agricultural Economics* 40(3): 281–294.

Mossie, M., Gerezgiher, A., Ayalew, Z. & Nigussie, Z. (2021). Welfare effects of small-scale farmers' participation in apple and mango value chains in Ethiopia. *Agrekon* 60(2): 192–208.

Murphy, S., Burch, D. & Clapp, J. (2012). Cereal Secrets: The world's largest grain traders and global agriculture. Oxfam Research Reports, August.

Mustonen-Park, H. (2018). Biotechnology Companies' Monopoly: Potentially Harming the Diversity and the Sustainability of Agriculture [Helsinki Metropolia University of Applied Sciences].

Neven, D., Odera, M.M., Reardon, T. & Wang, H. (2009). Kenyan supermarkets, emerging middle-Class horticultural farmers, and employment impacts on the rural poor. *World Development* 37(11): 1802–1811.

Nocke, V. & Whinston, M.D. (2022). Concentration thresholds for horizontal mergers. *American Economic Review* 112(6): 1915-1948.

Noton, C. & Elberg, A. (2013). Revealing bargaining power through actual wholesale prices. Documentos de Trabajo N°304, Serie Economía, Centro de Economía Aplicada, Universidad de Chile.

O'Brien, D. P. (2017). Price-Concentration Analysis: Ending the Myth, and Moving Forward. SSRN. Available at: <u>https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3008326</u>.

O'Connor, S. (2021). The hidden costs of powerful buyers and cheap prices. Opinion, Financial Times, Sept. 21, 2021. Available at: <u>https://www.ft.com/content/f65a5522-a7ac-436b-9100-a301111b19de</u>

Ochieng, D.O. (2017). Supermarket Contracts, Income, and Changing Diets of Farm Households: Panel Data Evidence from Kenya. *Food Discussion Papers 104*.

OECD (2014). OECD Competition Policy Roundtables: Competition Issues in the Food Chain Industry. Available at:

https://www.oecd.org/daf/competition/CompetitionIssuesintheFoodChainIndustry.pdf

OECD (2018a). Concentration in seed markets: Potential effects and policy responses, OECD Publishing, Paris. Available at: <u>https://doi.org/10.1787/9789264308367-en</u>.

OECD (2018b). Market Studies Guide for Competition Authorities. OECD Publishing, Paris. Available at:

www.oecd.org/daf/competition/market-studies-guide-for-competition-authorities.htm.

OECD (2020). Sustainability and Competition, OECD Competition Committee Discussion Paper. Available at: <u>http://www.oecd.org/daf/competition/sustainability-and-competition-2020.pdf</u>.

Onyeneke, R. U., Emenekwe, C. C., Chidiebere-Mark, N.M., Munonye, J.O., Aligbe, J. O., Kanu, C., Izuogu, C.U., Njoku, C.L., Uwazie, U.I., Uwadoka, C.O. & Azuamairo, G.C. (2020). Impact of Poultry Farmers' Participation in Modern Food Retail Markets on Household Dietary Diversity: Lessons from Southeast Nigeria. *Animals: An Open Access Journal from MDPI* 10(4): 611.

Parfitt, J., Barthel, M. & MacNaughton, S. (2010). Food waste within food supply chains: quantification and potential for change to 2050. *Philosophical Transactions of the Royal Society B: Biological Sciences* 365(1554): 3065–3081.

Perekhozhuk, O., Glauben T., Grings, M. & Teuber, R. (2017). Approaches and methods for the econometric analysis of market power: A survey and empirical comparison. *Journal of Economic Surveys* 31(1): 303–325.

Piccione, M. & Spiegler, R. (2012). Price competition under limited comparability. *Quarterly Journal of Economics* 127(1): 97–135.

Popkin, B. (2014). Nutrition, agriculture and the global food system in low and middle income countries. *Food Policy* 47: 91–96.

Poore, J., & Nemecek, T. (2018). Reducing food's environmental impacts through producers and consumers. *Science* 360(6392): 987–992.

Qaim, M. (2017). Globalisation of agrifood systems and sustainable nutrition. *Proceedings of the Nutrition Society* 76(1): 12–21.

Rao, E.J.O., Brümmer, B. & Qaim, M. (2012). Farmer participation in supermarket channels, production technology, and efficiency: The case of vegetables in Kenya. *American Journal of Agricultural Economics* 94(4): 891–912.

Rao, E.J.O. & Qaim, M. (2013). Supermarkets and agricultural labor demand in Kenya: A gendered perspective. *Food Policy* 38(1): 165–176.

Rao, E. J. O. & Qaim, M. (2011). Supermarkets, farm household income, and poverty: Insights from Kenya. *World Development* 39(5): 784–796.

Reardon, T., Barrett, C.B., Berdegué, J.A. & Swinnen, J.F.M. (2009). Agrifood Industry Transformation and Small Farmers in Developing Countries. *World Development* 37(11): 1717–1727.

Reardon, T. & Timmer, C.P. (2007). Transformation of markets for agricultural output in developing countries since 1950: How has thinking changed? Chapter 55 in R.E. Evenson and P. Pingali (eds.), *Handbook of Agricultural Economics, 3: Agricultural Development: Farmers, Farm Production and Farm Markets*, 2808–2855, Amsterdam: Elsevier Press.

Reardon, T., Timmer, C.P., Barrett, C.B. & Berdegue, J. (2003). The Rise of Supermarkets in Africa, Asia, and Latin America. *American Journal of Agricultural Economics* 85(5) – 1140–1146.

Reardon, T., Tschirley, D., Liverpool-Tasie, L. S. O., Awokuse, T., Fanzo, J., Minten, B., Vos, R., Dolislager, M., Sauer, C., Dhar, R., Vargas, C., Lartey, A., Raza, A. & Popkin, B. (2021). The processed food revolution in African food systems and the double burden of malnutrition. *Global Food Security* 28: 100466.

Reed, T. (2022). Using Randomized Experiments to Learn About Market Competition. *World Bank Blogs*, May 30, 2022.

Renko, S., Rai, S. & Kneevi, B. (2010). Environmental Responsibility of the Croatian Retailing. *International Journal of Management Cases* 12(2), doi: 10.5848/APBJ.2010.00096.

Rhodes, A. (2015). Multiproduct retailing. The Review of Economic Studies 82: 360–390.

Richards, T.J., Bonnet, C. & Bouamra-Mechemache, Z. (2018a). Complementarity and bargaining power. *European Review of Agricultural Economics* 45(3): 297–331.

Richards, T. J., Hamilton, S.F., & Yonezawa, K. (2018b). Retail market power in a shopping basket model of supermarket competition. *Journal of Retailing* 94(3): 328-342.

Rischke, R., Kimenju, S.C., Klasen S. & Qaim, M. (2015). Supermarkets and food consumption patterns: The case of small towns in Kenya. *Food Policy* 52: 9–21.

Robles, M., Torero, M. & von Braun, J. (2009). When speculation matters. IFPRI Issue Brief 57. February. Available at: <u>https://www.ifpri.org/publication/when-speculation-matters</u>.

Rogus, S., Athens, J., Cantor, J. & Elbel, B. (2018). Measuring micro-level effects of a new supermarket: Do residents within 0.5 mile have improved dietary behaviors? *Journal of the Academy of Nutrition and Dietetics* 118(6):1037–1046.

Rude, J., Harrison, D. & Carlberg, J. (2011). Market Power in Canadian Beef Packing. *Canadian Journal of Agricultural Economics/Revue Canadienne d'agroeconomie* 59(3): 321–336.

Ruel, M.T., Leroy, J.L., Ecker, O., Hernandez, M.A., Resnick, D. & Thurlow, J. (2020). Urban food systems and diets, nutrition, and health of the poor: Challenges, opportunities, and research gaps (with). Chapter 19 in J. Crush, B. Frayne and G. Haysom (eds.), *Handbook on Urban Food Security in the Global South*, Edward Elgar Publishing.

Russo (ed.), C. (2020). Pass-through of unfair trading practices in EU food supply chains. *Publications Office of the European Union*, Luxembourg. Available at: <u>https://publications.jrc.ec.europa.eu/repository/handle/JRC120994</u>.

Saitone, T.L. & Sexton, R.J. (2017). Concentration and consolidation in the U.S. food supply chain: The latest evidence and implications for consumers, farmers, and policymakers. *Economic Review (Federal Reserve Bank of Kansas City)* 102: 25–59.

Sexton, R. (2017). Unfair Trade Practices in the Food Supply Chain: Defining the Problem and the Policy Issues, in *Unfair Trading Practices in the Food Supply Chain: A Literature Review on Methodologies, Impacts and Regulatory Aspects*, Di Marcantonio, F. & Ciaian P. (eds.).

Sexton, R. & Xia, T. (2018). Increasing concentration in the agricultural supply chain: implications for market power and sector performance. *Annual Review of Resource Economics* 10(1): 229–251.

Sheldon, I. M. (2018). Industrial organization of the food industry. In *The Routledge Handbook of Agricultural Economics*, edited by G. Cramer, K. Paudel and A. Schmitz.

Sheldon, I. M. (2017). The competitiveness of agricultural product and input markets: A review and synthesis of recent research. *Journal of Agricultural and Applied Economics* 49: 1–44.

Shelke, K., Van Wart, J. & Francis, C. (2011). Social aspects of the food supply chain. In C. Baldwin, *Sustainability in the Food Industry* (pp. 145-159).

Shen, X., Zhang, L. & Zhang, J. (2021). Ratoon rice production in central China: Environmental sustainability and food production. *Science of The Total Environment* 764: 142850.

Shepherd, W.G. & Shepherd, J.M. (2004). The Economics of Industrial Organization. 5th ed. Long Grove, IL: Waveland Press.

Smith, G. (2009). Interaction of Public and Private Standards in the Food Chain. OECD Food, Agriculture and Fisheries Papers N°15. Available at: <u>https://doi.org/10.1787/221282527214</u>.

Smith, H. & Thomassen, O. (2012). Multi-category demand and supermarket pricing. *International Journal of Industrial Organization* 30: 309–314.

Solberg, S.O. & Breian, L. (2015). Commercial cultivars and farmers' access to crop diversity: A case study from the Nordic region. *Agricultural and Food Science* 24(2): 150–163.

Swinnen, J. (2020). Competition, market power, surplus creation and rent distribution in agri-food value chains. Background paper for The State of Agricultural Commodity Markets (SOCO) 2020. Rome, FAO. Retrieved from: <u>https://www.fao.org/documents/card/en/c/cb0893en/</u>.

Swinnen, J., & Vandeplas, A. (2015). Price Transmission in Modern Agricultural Value Chains: Some Conceptual Issues. In *Food Price Dynamics and Price Adjustment in the EU* (pp. 147–166). Oxford University Press.

Swinnen, J. & Vandeplas, A. (2010). Market power and rents in global supply chains. *Agricultural Economics* 41(S1), 109–120.

T4. (2021). Fast Food Market Share. Available at: <u>https://www.t4.ai/industry/fast-food-market-share</u>.

Tessier, S., Traissac, P., Maire, B., Bricas, N., Eymard-Duvernay, S., El Ati, J. & Delpeuch, F. (2008). Regular users of supermarkets in Greater Tunis have a slightly improved diet quality. *Journal of Nutrition* 138(4): 768–774.

Tubiello, F.N., Rosenzweig, C., Conchedda, G., Karl, K., Gütschow, J., Xueyao, P., Obli-Laryea, G., De Barros, J., Flammini, A., Mencos-Contreras, E., Souza, L., Quadrelli, R., Halldórudóttir Heiðarsdóttir H., Benoit, P., Hayek, M. & Sandalow, D. (2021). Greenhouse gas emissions from food systems: building the evidence base. *Environmental Research Letters* 16(6): 65007.

Umberger, W.J., He, X., Minot, N. & Toiba, H. (2015). Examining the relationship between the use of supermarkets and over-nutrition in Indonesia. *American Journal of Agricultural Economics* 97(2): 510–525.

UNEP (2021). Food Waste Index Report 2021. Nairobi. Available at: https://www.unep.org/resources/report/unep-food-waste-index-report-2021.

US Department of Justice. (2012). Competition and Agriculture: Voices from the Workshops on Agriculture and Antitrust Enforcement in our 21st Century Economy and Thoughts on the Way Forward. Available at: <u>http://www.justice.gov/ag/speeches/2010/ag-speech-100312.html</u>.

US Department of Justice. (2010). Horizontal Merger Guidelines. Available at: <u>https://www.justice.gov/atr/horizontal-merger-guidelines-08192010</u>.

Watson, E. (2016). Hartman Group to Supermarkets: Embrace Fresh or Die (Slowly). FoodNavigator-USA.com, August 10, 2016. Available at: <u>https://www.foodnavigator-usa.com/Article/2016/08/10/Hartman-Group-to-supermarkets-Embrace-fresh-or-die-slowly</u>

Wohlgenant, M. (2013). Competition in the US Meatpacking Industry. *Annual Review of Resource Economics* Vol 5(1): 1–12.

Zhou, Q., Zhao, L., Zhang, L., Xiao, Q., Wu, T., Visscher, T., Zhao, J., Xin, J., Yu, X., Xue, H., Li, H., Pan, J. & Jia, P. (2021). Neighborhood supermarket access and childhood obesity: A systematic review. *Obesity Reviews*, February 22 Suppl 1 (Suppl 1): e12937. doi: 10.1111/obr.12937.





Note: The arrows on the right-hand-side of the figure depict recent concentration trends along the different segments of the agri-food industry: a single arrow pointing upwards implies increasing concentration over the past years, while an arrow pointing upwards and another pointing to the right implies increasing concentration in some segments and stable concentration in others.

Source: Authors' own elaboration.





Note: Calculations based on adding company shares within each country for annual production turnover of fertilizers and nitrogen compounds at current prices (in US dollars applying year-over-year exchange rates).



Figure 2.2. Top 4 concentration (%) of pesticides and other agrochemical products in main producing countries, 2012 and 2021

Note: Calculations based on adding company shares within each country for annual production turnover of pesticides and other agrochemical products at current prices (in US dollars applying year-over-year exchange rates).



Figure 2.3. Top 4 concentration (%) of dairy processors in selected countries, 2012-2021

Note: Calculations based on adding company shares within each country for annual sales at retail selling price (in US dollars applying year-over-year exchange rates).



Figure 2.4. Top 4 concentration (%) of grain mill products in selected countries, 2012-2021

Note: Calculations based on adding company shares within each country for annual production at manufacturer selling price (in US dollars applying year-over-year exchange rates).



Figure 2.5. Top 4 concentration (%) of meat and meat products in selected countries, 2012-2021

Note: Calculations based on adding company shares within each country for annual production at manufacturer selling price (in US dollars applying year-over-year exchange rates).



Figure 2.6. Top 4 concentration (%) of grocery retailers in selected countries, 2012-2021

Note: Calculations based on adding company shares within each country for annual sales at retail selling price (in US dollars applying year-over-year exchange rates).

Source: Euromonitor International.

100



Figure 2.7. Top 4 concentration (%) of supermarkets in selected countries, 2012-2021

Note: Calculations based on adding company shares within each country for annual sales at retail selling price (in US dollars applying year-over-year exchange rates).

Level	Industry/category	Figure/Estimate	Period	Country/Market	Source and/or Reference
Upstream	Fertilizers (ammonia, urea, monoammonium phosphate (MAP), diammonium phosphate (DAP), nitrogen-	Top 5 countries control more than 50% of world's production capacity in ammonia, urea, monoammonium phosphate (MAP), diammonium phosphate (DAP), nitrogen-phosphate-potassium (NPK), phosphoric acid, and potash. China, India, Russia, and the US among the countries that dominate production capacity in the case of ammonia, urea, DAP/MAP, and phosphoric acid. Russia and Canada control more than 50% of world's production capacity of potash.	2008- 2009	Global	International Fertilizer Development Center (IFDC), Hernández and Torero (2011, 2013)
	phosphate-potassium (NPK), phosphoric acid, potash)	Top four firms control more than half of the production capacity of each major country producer. Concentration of potash production at the country level is the most extreme case: in four of the five main potash-producing countries, the CR4 index is 100%. For DAP/MAP and NPK, four of the five main producing countries exhibit a CR4 over 60%, while for urea three countries show a CR4 over 50%.	2008- 2009	Country level (main producing countries)	International Fertilizer Development Center (IFDC), Hernández and Torero (2011, 2013)
	Fertilizers and nitrogen compounds: nitrogen products associated with fertilizer production (nitric and	A global CR4 index of 56% is obtained for 2012 (compared to 43% in 1997) with China being the main producer (38%), followed by Russia (6%), India (6%), and Brazil (6%). The global level of concentration peaked in 2015 to 65% and has returned to 56% in 2021 where China (40%), Russia (6%), India (6%), and Brazil (4%) remain as the main producers.	1997- 2012- 2021	Global	Euromonitor International
	production (nitric and sulphonitric acids, ammonia, ammonium chloride, ammonium carbonate, nitrites, and nitrates of potassium) and phosphate, potash, and compound fertilizers.	At the country level, the CR4 indexes among the top four producing countries have similarly either maintained or slightly increased between 2012 and 2021. In India and Russia, the CR4 index is above 40%, while in Brazil and China it increased from 24.5% and 11.5% to 27.6% and 14%, respectively. The same companies also remain among the top producers in each company and in the same ranking; the only exception is Yunnan Yuntianhua Co Ltd in China that was not among the top four producers in 2012 and became the most important producer in 2021 with a 4.2% of the total volume of sales.	2012- 2021	Country level (main producing countries)	Euromonitor International
	Agrochemicals	Global CR4 index increased from almost 29% in 1994 to 53% in 2009.	1994- 2009	Global	ERS-USDA, Fuglie et al. (2011)

Table 2.1. Summary of figures and estimates of concentration by segment and industry of the agri-food chain

Level	Industry/category	Figure/Estimate	Period	Country/Market	Source and/or Reference
Upstream	Agrochemicals	Syngenta (Switzerland), Bayer (Germany), BASF (Germany), DuPont (USA), Monsanto (USA), and Dow (USA), the 'Big Six', are linked to the proprietary seed industry, and controlled 75% of the global pesticides market.	2014	Global	Action Group on Erosion, Technology, and Concentration (ETC), IPES- Food (2017)
	Pesticides and other agrochemical comprise insecticides and herbicides, plant growth regulators and disinfectants (orti growting	Compared to a global CR4 index of 52% in 1997 with the US being the main producer (27%) followed by Japan (9%), France (8%), and China (8%), in 2012 the concentration ratio increased to 64% with a substantial growth from China that concentrated 31% of the global production, followed by the US (16%), Brazil (10%), and Germany (7%). The level of concentration then peaked in 2015 with a global CR4 of 68% and has remained at 63% in 2021 where China (34%), US (14%), Brazil (8%), and Germany (7%) continue to be main producers.	1997- 2012- 2021	Global	Euromonitor International
	products), and fungicides and rodenticides.	At the country level, CR4 index in China increased between 2012 and 2021 (from 18% to 35.8%), where Wuhai He Ye Chemical Engineering Co Ltd has continued to consolidate its position as the major market player. In the US and Germany, in contrast, the concentration ratio decreased but still remain over 40%, while in Brazil it has remained around 9-10%.	2012- 2021	Country level (main producing countries)	Euromonitor International
		Market share of the four largest firms in the global seeds market increased from 21% in 1994 to 54% in 2009, based on firm sales data.	1994- 2009	Global	ERS-USDA, Fuglie et al. (2011)
	Seeds (crop seeds and biotechnology)	Global CR4 index of 58% in 2011 using sales data, which further increased to 65% in 2014. Bayer CropScience-Monsanto (Germany-US) explains 30.1% of the global sales followed by DuPont-Dow (US), Syngenta (Switzerland), and Vilmorin & Cie (France) with 22.7%, 7.8%, and 4.4%, respectively.	2011- 2014	Global	Action Group on Erosion, Technology, and Concentration (ETC), IPES- Food (2017)
	Maize	In maize, the CR4 index is 83%, where Monsanto alone has a 35.5% share and DuPont Pioneer another 34.5%.	2014	US	Begemann, S. (2015), Bryant et al. (2016)
	Soybeans	In soybeans, the CR4 is 76.2% where DuPont Pioneer and Monsanto have shares of 33% and 28%, respectively.	2014	US	Begemann, S. (2015), Bryant et al. (2016)

Level	Industry/category	Figure/Estimate	Period	Country/Market	Source and/or Reference
Upstream	Cotton	In cotton (upland) the CR4 is 91.4% where Bayer has a market share of 38.5% followed by Monsanto with a 31.2% share.	2015	US	AMS-USDA (2015), Bryant et al. (2016)
	Maize	In maize, the CR4 index is above 60% in 30 of the 32 countries covered and above 40% in the other two countries, while the HHI is above 1,500 in 27 of them; if the indicators are based on volume (instead of value), CR4 is over 60% in 28 countries and HHI is above 1,500 in 24 countries.	2016	Country level (32 countries, 65% of the global market value)	Kleffmann amis® AgriGlobe® database, OECD (2018a)
	Soybeans	For soybeans, the CR4 index is 42% in Ukraine while in the other six countries is above 60% in terms of both value and volume; the HHI is similarly over 1,500 in six of them, while in four countries is above 2,500.	2016	Country level (7 countries, 86% of the global market value)	Kleffmann amis® AgriGlobe® database, OECD (2018a)
	Rapeseed	For rapeseed, CR4 is above 60% in terms of value in 16 of the 17 countries and above 40% in the remaining country, while the HHI is over 1,500 in 12 countries; in terms of volume, CR4 is above 60% in 12 countries and the HHI above 1,500 in ten countries.	2016	Country level (17 countries, 62% of the global market value)	Kleffmann amis® AgriGlobe® database, OECD (2018a)
	Sunflower	For sunflower the CR4 index is above 60% across all countries in terms of value and only one country (Russia) has a CR4 below 60% in terms of volume (58%); the HHI in this market exhibits values over 2,500 across most countries.	2016	Country level (11 countries, 75% of the global market value)	Kleffmann amis® AgriGlobe® database, OECD (2018a)
Midstream	Livestock slaughter (steers and heifers; cows and bulls; hogs)	The share of steers and heifers slaughtered by the four largest firms increased from less than 40% in 1980 to 85% in 2015; for cows and bulls, from around 10% to close to 60%; and for hogs, from one-third to around two-thirds in the same period.	1980- 2015	US	GIPSA-USDA (2016), Deconinck (2021)
	Processed food (37 food manufacturing industries)	In the 37 food manufacturing industries covered by the US Census data in 2012, 29 were unconcentrated (HHI below 1,500); five were moderately concentrated (HHI between 1,500 and 2,500); and three (specialty canning, flavoring syrup and concentrate manufacturing, and other snack food manufacturing) were highly concentrated (HHI above 2,500).	2012	US	US Economic Census, Saitone and Sexton (2017)
	Processed food (42 food manufacturing industries)	In the 42 food manufacturing industries covered by the US Census data in 2017, 23 showed a CR4 index above 40% and 11 above 60%, some of which are: other snack food manufacturing; animal (except poultry) slaughtering; specialty canning; soybean and other oilseed processing; cane sugar manufacturing; and breakfast cereal manufacturing. Only 35 industries report HHI data, and 5 showed a HHI above 1,500 where the breakfast cereal industry the highest (2,210).	2017	US	US Economic Census, Crespi and MacDonald (2022)

Level	Industry/category	Figure/Estimate	Period	Country/Market	Source and/or Reference
Midstream	Processed food (23 food manufacturing industries)	Data for 14 member states in 2012 across 23 product categories, show that baby food, frozen ready-cooked meals, and cereals and coffee exhibit the highest concentration levels, while ham, bread, and cheese exhibit the lowest levels. When averaging across the 23 product categories in each country, the supplier concentration varies from a HHI less than 1,400 in Germany to over 2,800 in Denmark, with a median of 2,100. In addition, average concentration increased between 2004 and 2012 across most countries.	2004- 2012	EU (14 member states)	European Commission (2014)
	Processed food	A 2015 study by the Federal Economic Competition Commission (COFECE) using economic census data in Mexico reports estimated HHI less than 1,500 for sugar cane, vegetable oils and fats, cookies and pastries, among others; moderately concentrated markets in beer, prepared foods other than frozen foods, and condiments and dressings; and highly concentrated markets in snacks, breakfast cereals, and industrial baking.	2015	Mexico	Federal Economic Competition Commission (COFECE) Mexico, Deconinck (2021)
	Dairy products (butter and spreads, cheese, drinking milk products, yoghurt and sour milk products, and other dairy)	Except for Brazil and Russia, the concentration ratios are above or close to 50% and have generally remained rather stable or increased over time. China shows an 8.5-percentage points (pp) increase (from 46.9% in 2012 to 55.4% in 2021), followed by South Africa with a 7pp increase (from 53% to 60%), Australia with a 5pp increase (from 58.3% to 63.3%), and Saudi Arabia with a 2.9pp increase (from 51.5% to 54.4%). In Germany the ratio has remained rather stable at 49-50% while in the US it decreased from 58.3% to 53.6%.	2012- 2021	Main country in each geographical region available	Euromonitor International
	Grain mill products (grain milling and breakfast cereals)	For this product category, only Australia and the US exhibit important levels of concentration. In the case of Australia, the CR4 maintained around 74-75% between 2012 and 2021 while for the US it increased from 50.9% to 56.5%. For the remaining countries (Turkey, Brazil, Russia, and China), the concentration levels are below 20% and have remained rather stable over the last decade.	2012- 2021	Main country in each geographical region available	Euromonitor International
	Meat and meat products (red meat, white meat, and red and white meat products)	Australia exhibits the highest level of concentration, although it decreased from a CR4 of 67.6% in 2012 to 57.3% in 2021, followed by Brazil with a relatively stable CR4 of 43-44%. The other countries show concentration ratios below 30% that have generally maintained steady over time, except for Saudi Arabia that increased from 20.6% to 27.8%. In the US and Germany, CR4 indexes situate at 27.8% and 23.6%, respectively, in 2021, while China barely has a concentration ratio of 10.7% and Russia of 4.6%.	2012- 2021	Main country in each geographical region available	Euromonitor International
Downstream	International traders (grain)	The four major corporations that produce, process, transport, finance, and trade food and agricultural commodities (known as the ABCD): ADM - Archer Daniels Midland (USA), Bunge (USA), Cargill (USA), and Louis	2011	Global	Murphy et al. (2012)

Level	Industry/category	Figure/Estimate	Period	Country/Market	Source and/or Reference
Downstream		Dreyfus Commodities (France), are estimated to account for up to 90% of global grain trade.			
	Food Service wholesalers	Two firms, Sysco and US Foods (formerly U.S. Foodservice), dominate this broadline distribution in the US with over 70% of the market.	2020	US	AMS-USDA (2022)
	Fast Food	Leading brand in fast food market is McDonald's with 21.4% of total revenues in 2018, followed by Starbucks with 7.5%, and KFC and Subway with 2.8% each.	2018	US	T4 (2021)
	Grocery retailers	In the US, the combined market share of the four largest grocery retailers increased from 14% in 1984 to 22% in 1994 and 55 % in 2001.	1984- 1994- 2001	US	Swinnen and Vandeplas (2010)
	Grocery retailers	Market share of the largest five food retailers in Europe reached over 80% in the 2000s.	2000	EU	European Commission (2014)
	Grocery retailers	Einarsson (2007) reports that by 2004 in the Nordic countries the market share of the largest three retailers in the grocery market was already 79.6% in Finland, 81% in Iceland, 82% in Norway, and 91.2% in Denmark and Sweden.	2004	Nordic countries	Einarsson (2007)
	Grocery retailers	US Census data up to 2013 show that the four leading grocery retailers have a combined share of close to 40% of the national market (compared to less than 20% in 1992).	1992- 2013	US	US Economic Census, Saitone and Sexton (2017)
	Grocery retailers	In the period 1990 to 2015, the average HHI for food retail increased from 2,071 to 3,075 in rural counties from 1,617 to 2,924 in non-metropolitan small urban counties.	1990- 2015	US	ERS-USDA and Çakir et al. (2020) cited in Crespi and Macdonald (2022)
	Grocery retailers	By 2012, the HHI varied from less than 1,200 in Italy to more than 3,900 in Finland, with a median of around 2,000. Yet, in contrast with concentration among food manufacturers, average retail concentration saw a decline between 2004 and 2012 in 16 of the 26 member states for which data are available, due to growth of retailers who had a small market share in 2004 or were not even present (such as discounters), but in the other ten member states retail concentration increased.	2004- 2012	EU (26 member states)	European Commission (2014)
	Grocery retailers (supermarkets, other grocery (except convenience) stores, warehouse clubs, and supercenters)	The share of food sales by the top four retailers have been continuously growing from 31% in 2012 to 34% in 2019.	2012- 2019	US	ERS-USDA (2021a)

Level	Industry/category	Figure/Estimate	Period	Country/Market	Source and/or Reference
Downstream	Grocery retailers (includes supermarkets, hypermarkets, convenience stores, discounters, and other grocery stores)	Euromonitor shows important (over 40%) and somewhat increasing concentration levels among several countries. Australia reports the largest CR4 in grocery retail (from 74.4% in 2012 to 78.6% in 2021), followed by South Africa (from 47.2% to 54.3%), France (around 50%), the US (around 40%), and Russia that reached 40.2% in 2021 (exhibiting a 23.8pp increase relative to 2012). Of the remaining countries, the CR4 increased from 32.5% to 35.8% in Mexico and from 14.1% to 19% in Saudi Arabia, while in China the concentration ratio remained at 3%.	2012- 2021	Main country in each geographical region available	Euromonitor International
	Supermarkets (only retail outlets selling groceries within a selling space of 400-2,500 square meters; excludes discounters, convenience stores, and independent grocery stores)	In Germany, Australia, and South Africa, the CR4 index reached 96%, 89.6%, and 81.5%, respectively, in 2021 reflecting increases of 4-7pp from 2012. In the US and Saudi Arabia, CR4 levels reached around 40% in 2021, compared to 31% in 2012, while in Brazil the concentration ratio has remained relatively stable at 40-41%. Russia also shows an important increase from 12.1% to 22.8%, whereas in China the level of concentration is barely 3%.	2012- 2021	Main country in each geographical region available	Euromonitor International

Note: For additional details and references refer to the main text.

Source: Authors' own compilation.

ALL IFPRI DISCUSSION PAPERS

All discussion papers are available here

They can be downloaded free of charge

INTERNATIONAL FOOD POLICY RESEARCH INSTITUTE www.ifpri.org

IFPRI HEADQUARTERS

1201 Eye Street, NW Washington, DC 20005 USA Tel.: +1-202-862-5600 Fax: +1-202-862-5606 Email: <u>ifpri@cgiar.org</u>